IDS WORKING PAPER Volume 2016 No 469

Informality and Inclusive Growth in Latin America: The Case of Colombia

Cristina Fernandez and Leonardo Villar June 2016



Informality and Inclusive Growth in Latin America: The Case of Colombia Cristina Fernandez and Leonardo Villar

First published by Practical Action Consulting: Fernández, C., Villar, L. (2016) *Informality and Inclusive Growth in Latin America: The Case of Colombia*, ELLA Programme Regional Evidence Papers, Fedesarrollo: Bogotá; Practical Action Consulting Latin America: Lima

Second publication: IDS Working Paper 469 © Institute of Development Studies 2016 ISSN: 2040-0209 ISBN: 978-1-78118-313-7



This is an Open Access paper distributed under the terms of the Creative Commons Attribution Non Commercial 4.0 International licence, which permits downloading and sharing provided the original authors and source are credited – but the work is not used for commercial purposes. http://creativecommons.org/licenses/by-nc/4.0/legalcode

Available from: Communications and Engagement Unit, Institute of Development Studies, Brighton BN1 9RE, UK Tel: +44 (0) 1273 915637 E-mail: bookshop@ids.ac.uk Web: www.ids.ac.uk/publications

IDS is a charitable company limited by guarantee and registered in England (No. 87733)

Informality and Inclusive Growth in Latin America: The Case of Colombia

Cristina Fernandez and Leonardo Villar

Summary

The purpose of this research is to understand the impact of informality on inclusive growth in Latin America, and particularly in Colombia. The relationship between informality and inclusive growth is represented by two different hypotheses. According to the traditional perspective, informality constrains inclusive growth by restricting growth of the formal sector, draining resources from the public treasury and supplying low quality jobs that are unable to provide adequate conditions for a better standard of living. However, from another point of view, informality provides a default alternative to unemployment, particularly amongst poorly educated individuals; a perspective that is rarely taken into account in Latin America. Drawing on the case study of Colombia as well as a review of literature on Latin America, one of the main conclusions of this paper is that both hypotheses hold, yet their relevance depends on labour market restrictions and the socio-economic characteristics of informal workers since the informal sector in Latin America demonstrates considerable heterogeneity.

Keywords: informality; inclusive growth; labour markers; payroll taxes; wedge tax; labour tax; Latin America; Colombia.

Cristina Fernandez is a Senior Researcher at Fedesarrollo, and has extensive international experience form working as a country economist for development institutions including CAF, IADB and IFC. At IFC she worked on financial markets and the risks faced by the main financial institutions of Brazil, Mexico and South Africa. Cristina has a BA and MA in Economics from the Universidad the los Andes and an Economics MA from NYU. She has also worked at the Colombian Central Bank as researcher and advisor to the Governor.

Leonardo Villar is a Colombian economist who has been the Executive Director of FEDESARROLLO since 2012. Previously, he was acting Chief Economist and Vice-President of Economic Development and Public Policies at the CAF-Development Bank of Latin America. Leonardo was a Member of the Board of Directors of the Colombian Central Bank from 1997 to 2009. He has also been Technical Deputy-Minister at the Colombian Ministry of Finance, Foreign Trade Advisor to the Colombian Government, Vice-president of Economic Research and Planning of the Colombian Foreign Trade Bank (BANCOLDEX) and Technical Vice-president of the Colombian Banking Association. He has taught at the Universidad de los Andes and several other Colombian universities and has published extensively in the areas of macroeconomics, monetary, financial and foreign exchange policy and foreign trade. He obtained a Magister in Economics from Universidad de Los Andes in 1983. Later he studied at the London School of Economics, where he received his MSc and pursued doctoral studies.

Contents

	Sum Ackr Exec	mary, key words, author notes nowledgements cutive summary	3 6 7
1.	Intro	oduction	9
2.	Rese	earch design and methods	11
3.	Info	rmality and inclusive growth in Latin America	13
	3.1	Informality may promote inclusive growth by acting as a buffer to unemployment and creating opportunities for vulnerable populations: a literature review	17
	3.2	Informality may constrain inclusive growth not only by providing a low standard of life but also by impacting growth and jobs in the formal sector: a literature review	20
	3.3	Informality and public policies aimed at improving labour market efficiency and inclusive growth: a literature review	23
4.	Info	rmality and inclusive growth in Colombia	26
	4.1	Analysis of the channels through which informality might promote inclusive growth in Colombia (hypothesis 1)	25
	4.2	Analysis of the channels through which informality might constrain inclusive growth in Colombia (hypothesis 2)	40
	4.3	Policy analysis in Colombia: the impact of lowering the payroll tax (hypothesis 3)	0
5.	Mair	n findings	48
6.	Con	clusions and policy recommendations	50
	Refe	erences	51

Tables Regional informality rates (2004 to 2010) Table 3.1 Drivers of informality in Colombia. Motivations of informal self-Table 4.1 employed workers by socio-demographic characteristics Rates and shares in the Colombia labour market Table 4.2 Table 4.3 Logit model. Dependent variable: probability of formality

T I I A A		~ 4
I able 4.4	I ransition matrix	34
Table 4.5	Transition matrices by gender	34
Table 4.6	Differences in income, stability and satisfaction in Colombia before	
	and after adjusting by observable characteristics	37
Table 4.7	Fitting the matching model (probability of formality)	38
Table 4.8	Differences in labour income after adjusting by observable	
	characteristics and size	39
Table 4.9	Results of the differences in differences methods (OLS)	43
Table 4.10	Impact of the lowering of payroll taxes. Results from the differences	
	in differences and matching methods	44
Table 4.11	Impact of the tax reform by population group	47

	0	
Table 4.11 Impact of the tax	reform by population gro	up

Graphs

Graph 3.1	Informality in Latin America based on the firm definition	13
Graph 3.2	Informality. Rates based on different measures	15
Graph 3.3	Informality and inclusive growth	16
Graph 3.4	Payroll taxes as a percentage of commercial profits in sub-Saharan	
	and Latin American countries (2014)	25
Graph 4.1	Informality by metropolitan area in Colombia	30
Graph 4.2	Informality and skin colour	31
Graph 4.3	Formality rate and output gap in Colombia. Formality rate and	
	commodity exports	36
Graph 4.4	Impacts of the tax reform on the treatment and control groups	32
Graph 4.5	Lorenz curves	46

14

28

29

33

Acknowledgements

We would like to thank Paulo Mauricio Sanchez for excellent research assistance, Ani Rudra Silwal for his technical advice and Francisco Fernandez and Juan Camilo Medellin for their support. We would also like to thank Practical Action Consulting and the Institute of Development Studies for the advice and guidance they provided, particularly, Mark Lewis, Shandana Mohmand, Peter Houtzager and Don Leonard. Copy editing by Rebecca Clements.

This paper was first published by Practical Action Consulting Latin America.

Executive summary

The purpose of this research is to understand the impact of informality on inclusive growth in Latin America, and particularly in Colombia. The relationship between informality and inclusive growth is represented by two different hypotheses. According to the traditional view, informality constrains inclusive growth by restricting growth of the formal sector, draining resources from the public treasury and supplying low quality jobs that are unable to provide adequate conditions for a better standard of living. However, from another point of view, informality is a default alternative to unemployment, particularly among poorly educated individuals; a perspective that is rarely taken into account in Latin America.

The case study of Colombia is relevant to the region, not only because Colombia demonstrates regional averages in terms of informality and inclusive growth, but also because the informal sector is highly heterogeneous and this allows for simultaneous testing of both hypotheses. Another interesting feature of Colombia is that it is one of the few countries that have implemented an active policy to reduce informality in recent years. As such, it is expected that the experience of Colombia will provide useful insight for countries facing formal labour market restrictions such as Argentina, Brazil and Costa Rica. Notwithstanding, the impact of any policy will vary depending on the specific characteristics of informality in each country.

The analysis of the relationship between informality and inclusive growth has been understudied in general, perhaps because of the multiple channels that intervene in this relationship, as well as the lack of a consensus around the definition of both variables. For the purpose of this paper, we understand informality as those jobs or firms that are not regulated and/or do not contribute to public funds; and inclusive growth as growth accompanied by poverty reductions and equal opportunities for all segments of the population. Given this concept is rather broad, we have defined a set of inclusive growth indicators relating to the labour market as: employment, labour earnings, income distribution across different population groups, labour satisfaction, job stability and productivity. Exploring the way these variables impact inclusive growth is beyond the scope of this project. However, we have included in the literature review an assessment of evidence concerning the macroeconomic impact of informality, given the amount of externalities that tend to arise from individual labour decisions.

One of the main conclusions of this paper is that an analysis of informality cannot be performed assuming that informal workers are a mass of individuals with similar characteristics. In fact, in Latin America informal workers range from poorly educated individuals including women heads of household and the older population, to highly educated young adults living in productive cities. This finding corroborates new literature which points to heterogeneity in labour markets in terms of the coexistence of three types of informal workers: the 'structural' informal worker, for whom informality represents a default option to unemployment due to poor education, skills or experience; the 'induced' informal worker who is segmented from the formal labour market due to high entry restrictions or to less obvious restrictions, such as segregation; and, finally, the 'voluntary' informal worker who chooses to be informal.

The relationship between informality and inclusive growth is very different between these three groups. In terms of 'structural informality', informality represents a default option to unemployment and, as such, it is difficult to argue against the idea that among this group informality promotes inclusive growth. In contrast, 'induced informality' may constrain inclusive growth because labour market barriers prevent a population group from obtaining higher benefits and work stability available in the formal market. And, finally, 'voluntary informality', might be viewed as beneficial at an individual level, but not at an aggregate level

because individuals and/or firms restrict themselves to operating in a less productive environment in order to obtain certain 'free-rider' benefits. In any case, it cannot be ignored that informality negatively impacts inclusive growth at a macroeconomic level. Existing literature supports the view that informality lowers productivity, erodes the tax base, reduces productivity, produces unfair competition with the formal sector and increases corruption in Latin America.

The present analysis leads to important policy recommendations that are relevant not only in Colombia but in most developing countries. Although this research does not analyse policies implemented to address informality in detail, some lessons can be drawn from the literature review relating to the appropriateness of different approaches for targeting different groups of informal workers. While some informal workers might benefit from the removal of barriers to formal employment or affirmative action, structural informality must be tackled with different kinds of policy, such as improving access to education and flexible pension systems. Similarly, in the case of voluntary informality, monitoring and controlling informal professionals might be effective, whereas the same approach applied to structural informality might produce a negative impact on inclusive growth.

In December 2012, the Colombian government reformed the tax law by reducing payroll taxes from 29.5 per cent to 16 per cent. Our estimations show that this reform led to a one-off significant but moderate reduction in informality among the target population of about 7 percentage points, after controlling for some observable and unobservable variables. However, some of this reduction was the result of formal workers with flexible work becoming fully formal workers earning a minimum wage. We also found that the reform was even more effective among those workers with secondary education or less, showing that the impact of policies oriented towards reducing barriers to informality is not restricted to the higher tier of informal workers. This experience is of value for countries with high payroll taxes as well as high and enforced minimum wages.

1. Introduction

Informality is usually considered as a constraint to inclusive growth because, independently of our preferences for big or small governments, there is always the need for a pool of resources to meet communal expenses. When half of the working population or firms do not contribute to these expenses, as is the case in some Latin American countries, major problems can arise. A large informal economy also implies lower productivity, more corruption and represents a source of low quality jobs that are unable to provide adequate conditions for a better standard of living. However, informal employment also plays a key role in absorbing the unemployed and as such it might increase inclusive growth. This is particularly true for those groups considered vulnerable in terms of gender, race, age and education.

The purpose of this research is to understand the impact of informality on inclusive growth in Latin America and particularly in Colombia. To date, this relationship has been understudied, perhaps because of the multiple factors that interact in this relationship, as well as the lack of consensus around the definition of both variables. For the purposes of this paper, we understand informality as those jobs or firms that are not regulated and/or do not contribute to the public funds; and inclusive growth as economic growth accompanied by poverty reductions and equal opportunities for all segments of the population. Since this concept of inclusive growth is rather broad, we have restricted our set of indicators to: employment, labour income, distribution of earnings across different groups of the population, labour satisfaction and stability, and labour productivity. Examining the way these variables impact inclusive growth is beyond the scope of this research. However, we have included in the literature review evidence concerning the macroeconomic impact of informality, given the amount of externalities that tend to arise from individual labour decisions.

In order to explore our central research question, we investigate three specific hypotheses. First, *informality may promote inclusive growth by acting as a buffer to unemployment and creating opportunities for vulnerable populations*. By providing incomes to those who might otherwise be unemployed, informality allows individuals to engage in the economy and potentially benefit from economic growth. This is relevant for the structurally unemployed - those who have little hope of accessing the formal labour market - and for the cyclically unemployed, who may enter the informal sector during economic downturns, particularly if unemployment benefits, or any other type of transfers that favour unemployment over informality, are low or non-existent.

Second, informality may constrain inclusive growth both at an individual and at a society level through lower quality employment and lower productivity, and also by impacting growth and jobs in the formal sector. By preventing an important group of the population from obtaining higher wages, benefits and work stability, informality constrains inclusive growth, provided that these factors are present in the formal sector. In addition, informality generates lower productivity, erodes the tax system, increases the incidence of corruption and operates as a source of unfair competition to formal firms. Similarly, informal workers act as 'free riders' by congesting and not paying for public services.

Third, facilitating transitions from unemployment to informal employment and from informality to formality promotes inclusive growth. Smoother transitions may help ensure a more efficient allocation of resources and greater benefits for workers, firms and society as a whole. In the case of Colombia, we will focus on estimating the impact of reducing payroll taxes on informality. This issue is highly relevant, not only for Colombian authorities, but also for other governments in Latin America and Africa that might consider following this road in order to reduce informality.

The two general perspectives about the relationship between informality and inclusive growth described at the outset closely relate to older discussions around integrated, segmented and induced informal markets. First, the general idea of segmented markets (Lewis 1954; Harris & Todaro 1970) relates to 'structural informality' wherein for a particular portion of the population, informality represents the only available alternative to unemployment. Second, the notion of integrated markets (Maloney 2004), relates to those individuals/firms that voluntarily decide to take a job/operate in the informal rather than formal market. Under this scenario individuals/firms may benefit from informality at an individual, but not at an aggregate level because individuals/firms are restricting themselves to operating in a less productive environment. Finally, the concept of 'induced informality' (De Soto 2000) relates to individuals prepared to operate in the formal sector but segregated from this market by high entry barriers that are either explicit in legislation or implicit as in the case of segregation. These barriers constrain inclusive growth by preventing workers from obtaining higher wages, benefits and work stability (Loayza 1997). There is also a macroeconomic cost to informality that applies to all types described above (Levy 2008; Perry 2008).

The type of informality that prevails in each country will vary and has important implications for policy making. For example, while 'induced informality' might be tackled by removing formal employment barriers or by implementing affirmative action policies and childcare programmes, 'structural informality' requires a different policy approach, such as improved education and flexible pension systems, among others. Similarly, in the case of 'voluntary informality', providing economic incentives to formalise and enforce formality might be effective, whereas the same policy applied to structural informality could have a negative impact on inclusive growth. In terms of our three specific hypotheses, removing entry barriers to the informal labour market could facilitate transition from unemployment to informality, and, likewise, removing entry barriers to the formal labour market could improve the transition from informality to formality, depending on context-specific constraints to inclusive growth.

We started our analysis expecting to be able to locate countries according to either of the two general hypotheses and to formulate policy recommendations accordingly. However, one of the main conclusions of this paper is that informality cannot be analysed based on the assumption that informal workers are a bulk mass of individuals with similar characteristics. Instead, the informal sector should be understood as involving a mixture of individuals with different characteristics who operate in diverse environments. This finding corroborates new studies that have also identified heterogeneity in informality labour markets (Perry 2008; Garcia 2014; Ulyssea 2013; Alcaraz et al. 2012).

The case of Colombia is relevant for identifying lessons for the Latin American region not only because Colombia demonstrates regional averages in terms of informality and inclusive growth, but also because the country is home to a highly heterogeneous informal sector and this allows for simultaneous testing of both hypotheses. Another interesting feature of Colombia is that it is one of the few countries that have implemented policies aimed at reducing informality in recent years. Thus it is expected that lessons drawn from Colombia will be useful to other countries facing high formal labour costs, such as Argentina, Brazil and Costa Rica. Notwithstanding, the impact of any policy addressing informality will vary depending on the particular characteristics found in each country.

This paper is structured as follows. Section 2 presents the research and design methods; section 3 presents evidence from the Latin America regional literature review; section 4 provides an analysis of the case of Colombia; section 5 summarises the main lessons, and section 6 provides conclusions and policy recommendations.

2. Research design and methods

The complexity of analysing the relationship between informality and inclusive growth is due in part to the lack of a common or standard definition for either of these concepts, the variety of channels that explain the impact of informality over inclusive growth and the number of variables that affect inclusive growth outside the labour market. We address these issues as follows.

Definition and measurement of informality

The definition of informality varies, however for the purpose of this paper we understand it as jobs or firms that are not regulated and/or do not contribute to public funds, as this covers the majority of definitions provided in the literature. Both firms and individuals are included in the definition, which means that informal workers can work either in the formal or the informal sector.

There are also several ways to measure informality. The two most commonly used are: (i) the firm definition, which is closely related to the type and size of the firm and therefore with the quality of the job,¹ and; (ii) the legal definition, which relates to compliance with legal requirements and the kind of social protection provided to the worker. The correlation coefficient for both measures in Latin America is 0.85 (Tornarolli et al. 2012). In the case of Colombia, we mostly use the firm measure because it is most frequently used by Colombian authorities and the most suitable for making international comparisons since it does not vary according to local enforcement and regulation policies (as the legal definition may). The firm definition also enables us to maintain greater consistency across the research since we have chosen to explore the impact of lowering payroll tax, and the legal definition is not amenable to the method that we selected. We occasionally do use the legal definition, however, to provide robustness to our findings or where necessary due to data availability.

Definition and measurement of inclusive growth and its relationship with informality

As with informality, there is no standard definition of inclusive growth. However, the literature broadly understands it as *growth accompanied by poverty reductions and equal opportunities for all segments of the population*. Attempts to measure inclusive growth include McKinley (2010), Ali and Son (2007), Almeida-Ramos, Ranieri and Lammes (2013) and Anand, Mishra and Peiris (2013).

Using these inclusive growth indexes, or any other welfare analysis methodology, to analyse informality and inclusive growth would necessarily imply an oversimplification of our findings and may compromise their applicability across different contexts. Instead, we use data-based evidence methodologies to review each of the channels through which informality may affect inclusive growth. By using this approach, rather than generating a 'one-size-fits-all' recommendation, we recognise the diversity in country contexts. In addition, we limit the set of variables for analysing inclusive growth to those related to the labour market, namely: employment, labour income, distribution of earnings across groups of the population, labour satisfaction and stability, and labour productivity. Exploring the way these variables impact inclusive growth is beyond the scope of this research. However, we do include a literature review of the macroeconomic impact of informality, given the amount of externalities that tend to arise from individual labour decisions.

¹ This category includes workers employed in firms with no more than five employees; unpaid family helpers or housekeepers; self-employed with the exception of independent professionals and technicians; and business owners or firms with no more than five workers. This criterion changed from 10 workers or less (ILO10) to 5 workers or less (ILO5) showing a higher correlation with other measures of informality (Bernal 2009).Since 1999 the Delhi Group established the ILO5 as the standard measurement of informality (Central Statistical Organisation 1999).

Channels through which informality impacts inclusive growth

As explained previously, our analysis is focused on some relevant channels that occur inside the labour markets. Channels selected for each hypothesis and the methods used to approach each of them are explained below.

Channels through which *the transition from unemployment to informality can improve inclusive growth* are: (i) informal employment reduces unemployment, particularly among some vulnerable groups; (ii) informal employment is associated with a higher income than unemployment, even in the presence of unemployment benefits; and (iii) informal employment can prove to be useful in economic downturns to prevent serious falls in living standards, however this depends on the informal sector constituting a counter-cyclical variable. The methods used to analyse these channels are: (i) review of surveys identifying informality as a default alternative to unemployment; (ii) characterisation of informal workers, and particularly the most vulnerable, to identify the incidence of informality among this segment; (iii) transition matrices to analyse the frequency of transitions between groups and whether these transitions are increasing inclusive growth by facilitating the functioning of the labour markets, and; (iv) pro-cyclical analysis of informality to confirm whether informality acts as a buffer to unemployment during hard times².

Channels through which *informality constrains inclusive growth* are: (i) informal employment generates less income, benefits and stability than formal employment; (ii) informal sector productivity is lower than formal sector productivity, controlling for observable variables such as size, and; (iii) informality erodes the tax base and/or the quantity and/or quality of public services. The methods used to analyse these channels are the following: (i) analysis of income, work satisfaction and stability of informal versus formal workers controlling for observable characteristics and using Matching methods; (ii) analysis of differences in productivity between formal and informal workers controlling for size and using Matching methods and (iii) analysis of the fiscal impact of informality based on a review of relevant literature.

Concerning *policies oriented at facilitating labour market transitions,* we concentrate our efforts on measuring the impact of lowering the payroll tax through a methodology that mixes Matching and Differences in Differences techniques. The reasons for applying this methodology instead of other techniques are explained in section 4. We also analysed the distributional impact of this particular reform by measuring its impact on specific segments of the population as well as Lorenz curves.

² The pro-cyclical analysis of informality also helps to identify whether informality is the result of an individual choice, which is explained later.

3. Informality and inclusive growth in Latin America

With the exception of Heintz (2012), who presents a general approach and formulates an agenda to analyse the relationship between informality and inclusive growth, this topic has not been explicitly researched at a global level or in the Latin American region. However, some literature does analyse the channels through which informality impacts inclusive growth. This evidence, organised according to our three hypotheses, allows us to get a good grasp of the mechanisms through which informality impacts inclusive growth in the region. In this section, we begin by presenting an overview of informality and inclusive growth in Latin America as well as the main attempts to measure this relationship. Second, we summarise the literature that explores the channels through which informality promotes and constrains inclusive growth. Third, we review policies aimed at increasing inclusive growth by tackling informality and, finally, we provide conclusions and identify the knowledge gaps that we intend to fill in the subsequent section.

Informality in Latin America and Colombia

Informality is a challenge of great dimensions in Latin America. The average informality rate among the largest 14 economies is 46 per cent, representing nearly 130 million informal workers in the region (ECLAC/ILO 2015). In countries like Bolivia, Honduras and Paraguay, up to 70 per cent of workers are informally employed (OECD 2015). Furthermore, informality rates are particularly high among vulnerable groups in the region: 75 per cent of all workers with low levels of education, 50 per cent of working women (FORLAC 2014) and 55 per cent of young workers are informal workers (FORLAC 2015).



Despite these figures, Latin America does not show particularly high informality rates among developing countries. According to WIEGO (2014), sub-Saharan Africa, South Asia and East Asia have even higher average informality rates as a percentage of non-agricultural employment. Nevertheless, as shown in Table 3.1, given the higher proportion of economically active population and urbanisation rates in Latin America, urban informality is higher relative to the size of the total population than it is in African countries, for example. Notwithstanding, informality rates have declined in Latin America over the last ten years. From the early nineties to the early 2000's, the informality rate remained constant at around 53 per cent and then lowered to 46.8 per cent in 2013, as shown in Graph 3.1. This period of improvements in the labour market coincided with high growth rates, and it is therefore difficult to establish the drivers behind this reduction. As such, the question arises whether or not informality may go back to its previous level as a result of the recent downturn in the region's Gross Domestic Product (GDP).

In Colombia, almost half of the working population is engaged in the informal sector, despite the fact that informality shows a declining trend. As of June 2015, 48 per cent of the working population in the 13 main metropolitan areas was informal, compared with 54.7 per cent back in 2002 (GEIH 2007-2015; ECH 2002-2006).³ In Graph 3.2 we compare this measurement with other measurements for labour informality, such as workers who do not make contributions to state health and pension schemes or do not contribute to either. As Graph 3.2 shows, although there are some differences in level of informality, which ranged between 44 per cent and 51 per cent in 2014 depending on the definition adopted, all the measures demonstrate similar behaviour across time, including a negative trend since the year 2000. Thus, the number of workers in the formal sector and people contributing to non-wage benefits has increased in the country.

Table 3.1Regional informality rates (2004 to 2010)							
	Informality/non agricultural employment	Non agricultural employment/population					
Latin America and the Caribbean	51	50					
Sub-Saharan Africa	66	28					
Middle East and North Africa	45	33					
Eastern Europe and Central Asia	10	46					
South Asia	82	26					
East and Southeast Asia (excluding China)	65	38					
China *	33	N/A					

Source: Wiego (2014) and own calculations

* Estimates for urban China based on six cities: Fuzhou, Guangzhou, Shanghai, Shenyang, Wuhan and Xi-an

Inclusive growth and informality in Latin America and Colombia

According to the inclusive growth index produced by Almeida-Ramos et al. (2013), which takes into account per capita income growth, income distribution, poverty reduction and employment generation in 2006, most Latin American countries are ranked between the high-medium range, such as Mexico and Uruguay; the medium range, including Brazil, Chile,

³ Based on ILO5.

Ecuador, El Salvador, Paraguay and Peru; and the medium-low range, such as Bolivia, Colombia, the Dominican Republic and Panama. Similarly, the index produced by Anand et al. (2013), which takes into account per capita income growth and income distribution variables from 1990 to the latest available data in 2013, classifies Venezuela, Bolivia and Colombia as countries with low levels of inclusive growth and Argentina, Brazil, Ecuador and Mexico as Latin American countries with medium levels of inclusive growth.



Adopting a less systematic approach, Cord (2015) claims that after the eighties and nineties when Latin America was last or second to last in terms of regional growth and income distribution, the region emerged with one of the best performances in terms of inclusive growth during the 2000's, registering a growth rate higher than the world average, reducing extreme poverty by half and achieving considerable reductions in income gaps. Similarly, Colombia, a country with low to medium-low inclusive growth levels, has recently achieved remarkable results reducing poverty from 50 per cent in 2002 to 28.5 per cent in 2014. Regarding income distribution, however, the results were less impressive. Although the Gini coefficient that measures income inequality reduced from 0.57 in 2002 to 0.53 in 2014, it continues to be one of the highest in the world. Neither of these indexes and analyses takes into account the most recent period, 2015 in particular, during which time growth has seriously slowed in most countries of the region, probably affecting these achievements.

Graph 3 plots the two reviewed indexes of inclusive growth against informality for the set of countries for which data is available⁴. It is possible to observe a slightly negative but low correlation between the two variables. The correlation coefficient between the level of self employment and the level of inclusiveness, according to the 2006 Almeida-Ramos et al. (2013) index⁵, is -0.02. The correlation coefficient in terms of changes, using the Anand et al. (2013) index and self-employment⁶, is -0.3. Both relationships are significant to 95 per cent. As can be observed in Graph 3.3, South Africa is an outlier in this matter, whereas Colombia, showing more standard results, is one of the countries with the most negative correlation between informality and inclusive growth.



and Anand et. al. (2013). Note: In the case of South Africa, change in self-employment is calculated with respect to the 2000's due to data availability.

Results in Graph 3.3 also suggest that the relationship between inclusive growth and informality is either negative or non-existent. A negative correlation is consistent with the literature that relates informality to some of the components commonly used to measure inclusive growth. Loayza and Rigolini (2006), for example, found that countries with low GDP per capita tend to have higher self-employment rates, used as a proxy for informality. They also found that an increase in one standard deviation in informality leads to a decline of

⁴ These countries are Argentina, Brazil, Bolivia, Botswana, Burkina Faso, Chile, Costa Rica, the Dominican Republic, Ecuador, El Salvador, Jamaica, Madagascar, Mexico, Namibia, Nicaragua, Paraguay, Panama, Peru, South Africa, Tanzania, Uruguay, Venezuela and Zambia.

⁵ Or the inverse index, since for Ramos et al. (2013) a higher index means less inclusiveness whereas for Anand et al. (2013) a higher index means greater inclusiveness.

⁶ We used self-employment as a proxy, since the ILO does not provide informality series.

0.7 to 1 percentage points in the rate of per capita growth and an increase of 0.12 to 0.24 percentage points in poverty rates⁷. Perry et al (2007) found a strong correlation between informality and inequality at a global level, while Maurizio (2013) and the Economic Commission for Latin America and the Caribbean (ECLAC 2014) found that formalisation had a positive effect on labour income distribution in Latin America. Nevertheless, since it is small, the correlation coefficient between informality and inclusive growth indexes might also suggest the existence of conflicting forces that underlie informality, and which may promote or restrict inclusive growth, as we have suggested in our two hypotheses.

3.1. Informality may promote inclusive growth by acting as a buffer to unemployment and creating opportunities for vulnerable populations: a literature review

This section provides a review of literature relating to our first hypothesis by examining surveys reports, transitions, counter-cyclicality movements of informal employment and incidence of informality among vulnerable groups.

In the previous sections, we hypothesised that informality might be positive for inclusive growth by offering a substitute to unemployment, provided that informal earnings are higher than unemployment benefits, which is not an issue in Latin America given the lack of these benefits. It should also be noted that in the case of 'induced informality', providing an alternative to unemployment can only result in more inclusive growth if reducing explicit or implicit barriers to the formal labour market is not an option.

Drivers of informality

From the point of view of preferences, Arias and Bustelo (2007) and Arias, Landa and Yañez (2007) use household surveys to determine that 59 per cent of the self-employed in Colombia would prefer to have a job in the formal market. This figure amounts to 40 per cent in Argentina, 26 per cent in Bolivia and 25 per cent in the Dominican Republic. Among informal salaried workers, preferences for a formal job are 60 per cent in Colombia, 57 per cent in Argentina, 48 per cent in Bolivia and 43 per cent in the Dominican Republic.

Although not strictly comparable with one another, the surveys also reported that a common reason for self-employment is the impossibility of finding a salaried job. This was cited as the main reason by 25 per cent of interviewees in Bolivia; the first or second most important reason by 55 per cent and 52 per cent of interviewees in Colombia and the Dominican Republic respectively, and as a relevant reason amongst 58.8 per cent of Argentineans⁸. According to Perry et al. (2008)⁹, in Mexico, 44 per cent of the male population chose self-employment because of higher earnings and only 12.4 per cent because they could not find a job in the formal market. Among women, only 6 per cent reported that they could not find a job in the formal market. In Brazil, 33 per cent of people in informal self-employment said they would leave their current job for a salaried contract. The surveys in Mexico and Brazil also show a higher preference for independent work among women and older populations. While these results demonstrate that an important portion of the population works informally because they can't find a job in the formal market, there is also clear and significant heterogeneity between and within countries, with Colombia showing the highest preference for formal employment and Mexico the lowest.

⁷ All the results were significant and controlled for enforcement and partially controlled for endogeneity through instrumental variables.

⁸ The survey asked for the main reason in Bolivia, up to two reasons in Colombia, and the two most important reasons in the Dominican Republic, whereas in Argentina the question permitted multiple responses.

⁹ Based on local surveys. In the case of Mexico, the survey asked respondents to identify the main motivation.

Counter-cyclicality

Another popular indicator used to measure whether informal employment is a substitute to unemployment is counter-cyclicality. According to Tornarolli et al. (2014), in the presence of labour market rigidities and involuntary informal sector employment, when the economy enters into recession and a minimum wage exists, some of the formal firms fire workers who subsequently find refuge in the informal sector. Therefore, the ratio of informal to formal workers tends to increase during downturns. Similarly, when the economy grows the cost of hiring becomes relatively lower and the ratio of informal to formal employment decreases (Loayza and Rigolini 2006). However, in the presence of voluntary informal workers, during upturns, an increase in the informal wage should attract informal workers and increase the size of the sector, parallel or even pro-cyclically to the increase in formal employment¹⁰.

Some of the most important findings in this respect can be summarised as follows. Bosch and Maloney (2008), in the case of Mexico and to a lesser extend in Brazil, found that transitions between formality and informality operate as though the two markets were well integrated, whereby informal employment does not behave as a substitute to unemployment in Mexico. Using information from household surveys, Tornarolli et al. (2014) found the informality rate to be counter-cyclical in Brazil, Colombia and Chile, mostly among salaried informal workers and in Uruguay and Venezuela, mostly among the self-employed, and did not find any clear relationship in other countries such as Costa Rica and Peru. Loayza et al. (2006) also found good evidence confirming the prevalence of counter-cyclicality of informality in the region. Finally, Fiess et al. (2008) found that whereas informal employment in Mexico and Brazil tends to be pro-cyclical, in Argentina and Colombia this relationship is counter-cyclical. Their main finding, however, is that the issue of counter/pro-cyclicality is very much associated with the shocks faced by the countries and therefore it is dependent on the period of analysis. In sum, most studies signal that informal markets tend to be procyclical in Mexico and counter-cyclical in Argentina, Chile, Colombia, Uruguay and Venezuela. Evidence for Brazil suggests pro-cyclicality, however results are less conclusive in this case.

Incidence of informality among workers with low education, skills and experience In general, there is a higher incidence of informality among certain vulnerable groups in Latin America, namely: women heads of household, the old and the very young and very loweducated workers. This would not be a problem if labour decisions were voluntary, in fact in this case informality might increase inclusive growth at an individual level. This is less clear in the case of workers that would prefer to have a job in the formal market but are unable to find it, either because they have low levels of education and experience or because they face high entry barriers to the formal market. Drawing on this approach, we first review literature that analyses the incidence of informality among the poorly educated population and the very young population. We then analyse the incidence of informality and signs of segmentation among the women and ethnic minorities.

According to ILO (2013a), the incidence of informality among the low educated population in Latin America is high, averaging 63 per cent among workers with primary education and 75 per cent among workers with less than primary education; compared with a rate of 47 per cent for those with secondary education. Similarly, Gong, Van Soest and Villagomez (2004) showed that in Mexico education is negatively correlated with informal work. Haanwinckel and Soares (2014) found in Brazilthat an increase in the ratio of skilled to unskilled workers between 2003 and 2012 was partly related to a decline in informalityand that without this increment informality would have increased by 4.8 percentage points, instead of declining by

10

According to Fiess, Fugazza and Maloney (2008), during booms generated by commodity exports, which promote the informal-intensive service sector, one might expect informal employment to be even more pro-cyclical than formal employment.

6.4. Finally, Herrera, Lopez and Montellon (2013) found that differences in income between formal and informal workers in Colombia are much higher for the low educated population because iinformal workers face not only lower returns on their education, but also suffer a second penalty associated with educational mismatches. However, as El Badaoui and Rebiére (2012) point out, even if higher education means a better chance of engaging in the formal sector, it does not mean that educated workers cannot search for an informal job.

Concerning age, the young population in Latin America demonstrates an average informality rate of 56 per cent compared with 46 per cent among the adult population. This segment also shows an unemployment rate two to four times higher than that of adults in the region (ILO 2013a). In the case of Mexico, Perry (2008) found that working in the informal sector was much more involuntary for the younger population than for adults, while this is not the case in Brazil. Bassi, Busso, Urzea and Vargas (2012) claim that one of the problems with unemployment among the young population is the low probability of them transitioning from an informal to a formal job; a figure that stands at just 10 per cent in Mexico, 15 per cent in Argentina and close to 30 per cent in Brazil and Chile. There is little evidence of informality rates amongst the older population in Latin America, with the exception of Fedesarrollo (2015) which shows a recent important increment among this group in the case of Colombia.

Incidence of informality among women and ethnic minorities

In Latin America, the incidence of informality is higher for women (50 per cent) than for men (45 per cent) (ILO 2013a). However, as explained previously, this does not mean much in terms of inclusive growth if it is the result of a choice. In fact, it would be perfectly understandable for a woman with children to choose informal work because of the flexibility it provides. The Mexican survey, and to a lesser extent in the Brazilian survey, women showed a relatively higher preference for informal work (Arias & Bustelo 2007; Arias, Landa & Yañez 2007). In the case of Colombia, there difference for independent workers is 5 per cent, after controlling for other observable characteristics (Bernal 2009). Likewise, Galli & Kucera (2008) found that in Latin America women's behaviour does not differ from men's in terms of counter/pro-cyclicality. Therefore, it is not easy to establish whether the incidence of informality among women is source of segregation or a matter of preference. In particular, there is a current lack of knowledge on this subject concerning women heads of household for whom the decision to work may be less of a choice.

Regarding race segregation, according to Henley, Arabsheibani and Carneiro (2006), in Brazil whites and Asians show the lowest rates of informality, whereas mixed races and blacks face the highest rates. Similarly, Tokman (2008) claims that countries with larger indigenous populations demonstrate higher informality rates. He also searched for bias in the labour market against immigrants, but found no support for his hypothesis. Bernal (2009) also reported a high incidence of informality among ethnic minorities in Colombia.

In sum, there is a high incidence of informality among vulnerable groups in Latin America who seem to perceive informality as a default option to unemployment. This is particularly true in the case of the poorly educated, the young, indigenous people and ethnic minorities. This finding is important in determining whether informality increases inclusive growth, since informality in vulnerable groups is likely to be understood as 'structural informality' or 'induced informality', if it is the result of segregation.

Informality as a default option to unemployment in Colombia

In Colombia, there is no consensus around whether the labour market is segmented or integrated. There is some evidence supporting the idea that integrated markets existed during the period 1991-1996 and became more segmented after the nineties, a decade characterised by increases in payroll taxes and low productivity (Peña 2013; Mondragon, Peña & Kugler 2010; Perry 2007; Maloney 2004). These findings corroborate Fiess et al. (2008), which shows a high, negative and significant rolling correlation between the

economic cycle and informality between1997 and 2002, and no significant correlation in subsequent periods. On the other hand, Pratap and Quintin (2004) were unable to find any strong evidence of segmentation between the formal and the informal sectors in Colombia. Recent literature (Garcia, 2014; Perry, 2008), which understands informality as a mixture of integrated and segmented markets, reflects the heterogeneity of the labour market in the country¹¹.

Summary

The question we were trying to address in this section was whether informality in Latin American countries can be positive for inclusive growth. We learned that if informality is a default and not a voluntary option, and if it is due to low productivity, it can even be positive for inclusive growth. In fact, according to Busso, Fazio and Levy (2012) informality employs an important segment of the population with low education and/or skills that otherwise would not find a job in the formal market because the cost of hiring them is greater than their productivity value for the firm. If the government enforces formality, this group will face unemployment and will probably end up worse off than when working informally. However, in the context of 'induced informality', where informal workers demonstrate similar productivity than formal workers, permitting informality as an alternative to unemployment only represents the second best policy option, the first being to reduce implicit or explicit formal market entry restrictions.

The distribution of the three types of informality in each country is not really covered by existing literature¹², perhaps because even at an individual level structural, induced and voluntary informality are not straightforward to identify. In fact, a worker can operate in all three types of informality at the same time. Despite this challenge, we were able to gather evidence to identify which type of informality prevails in each country. The evidence indicates significant heterogeneity across Latin America, not only within countries but also between countries and macroeconomic contexts. This finding corroborates recent literature that posits informality as a dynamic combination of all three types of informality (Perry et al. 2007; Fies et al. 2008; Ulyssea 2013). Despite this uncertainty, and at the risk of over-simplifying, one could argue that whereas Mexico shows a higher prevalence of 'voluntary informality', Colombia seems to have a very heterogeneous mixture of informality.

3.2. Informality may constrain inclusive growth not only by providing a low standard of life but also by impacting growth and jobs in the formal sector: a literature review

The channels through which informality might constrain inclusive growth are: (i) informal employment generates less income, benefits and stability than formal employment; (ii) informal sector productivity is lower than formal sector productivity controlled for observable characteristics as size, and; (iii) informality erodes the tax base and/or the quantity and/or quality of public services. The findings of the previous section also apply here since informality is understood as a mixture of 'structural informality', where the first hypothesis – informality increases inclusive growth -is more relevant, and 'induced' and 'voluntary' informality where the second hypothesis – informality constrains inclusive growth -is more relevant.

¹¹ Garcia (2014) used wages as a criterion for segmentation inside the labour market and found that big cities in Colombia tend to show a larger proportion of voluntary informality. However, Perry (2008) argues that wages do not necessarily demonstrate segmentation but can in fact signal differences in unobservable characteristics among workers, as well as preference for informality.

¹² One exception is Alcaraz, Chiquiarand Salcedo (2012) who found that between 10 and 20per cent of informal workers in Mexico demonstrated marked signs of segmentation based on the personal characteristics of individuals.

Informal employment generates less income, benefits and stability than formal employment

In general, informal workers are at a disadvantage in terms of earnings when compared with formal workers. According to the ILO (2013a), the incidence of informality in Latin America reduces from 72 per cent among the lowest earning quintile to 31 per cent among the highest earning quintile. Similarly, the incidence of poverty among informal workers is between two and five times higher than that of formal workers (Maurizio 2015).

However, this income gap might only reflect different worker productivities due to differences between the two sectors in terms of observable characteristics such as education or unobservable characteristics such as social skills. Therefore, wage differences between formal and informal workers should be controlled by observable characteristics for specific sectors of the population or using more sophisticated techniques to control by unobservable variables. Controlling by differences in preference for informal work, another possible cause of wage differentials, is a more difficult task.

Maurizio (2015) found a positive earning gap in favour of formal jobs in the cases of Argentina, Brazil, Chile and Peru. This gap tends to decrease with the level of income and becomes negative at high-income levels in the case of Chile and Brazil. Similarly, Bargain (2010) found that low-income informal self-employed workers face lower earnings than their formal sector peers, whereas workers in the high-earning tier of informal self-employment receive a significant earning premium, as is the case in Brazil, Mexico and South Africa. Perry et al (2008) performed a similar exercise for Argentina, Bolivia and the Dominican Republic also finding a positive gap for low-income workers in the cases of Argentina and Brazil and a negative gap for high-income workers in Bolivia and high-income self-employed workers in Argentina. As for Colombia, Herrera, Lopez & Montellon (2013) found a positive income gap in favour of the formal sector for all income quintiles. Also in the case of Colombia, Daza and Gamboa (2008) found a gap in favour of formal market earnings ranging between 30 per cent and 60 per cent depending on the definition of informality used, with small variations over the period 2008 to 2012.

Other than income, there are other advantages to being formally employed versus informality, including access to social protection in terms of health benefits and old age pension as well as job stability. Even if a worker does not value these perks, for society as a whole it is important that workers and their families are protected against these risks. A similar argument can be applied to worker training. According to Perry et al (2007), formal workers tend to be more satisfied with their jobs in Colombia, however this is not necessarily the case in the Dominican Republic or Argentina where formal and informal workers report similar levels of welfare.

Informal sector productivity is lower than formal sector productivity controlled by observable characteristics

One of the most powerful arguments against informality is the impact on productivity. According to Levy (2008), formal market restrictions are associated with a misallocation of firms in favour of small firms that operate more intensively in the informal sector. This occurs not only because there is an implicit subsidy to informality but also because informal firms try to keep themselves out of the sight of authorities in order to continue being informal. According to Levy, the low productivity embodied by informality is not caused by the existence of self-employment or family firms, but rather it is due to the existence of *too many* of them. In fact, a small amount of self-employed workers and family firms can have a positive impact on informality by completing market functioning.

However, size is not the only cause for differences in productivity. Other factors affecting the informal sector include limited access to credit (Dabla-Norris & Koeda 2008), fewer incentives to train and adopt technology (Dabla-Norris, Gradstein & Inchauste 2007), a lower

probability of engaging in the export market and restricted access to public goods, such as the legal system and government labour training schemes. Ydrovo (2010) even suggests that informality affects aggregate productivity by weakening the process of creative destruction.

In the case of Mexico, Busso, Fazio and Levy (2012) found that one peso invested in capital and labour within formal firms is worth 28 per cent more than if it is invested in illegal firms, and 50 per cent more than if it is invested in legal and informal firms¹³. They also found a difference of productivity of 84 per cent in favour of formal firms, controlled by observable characteristics. In the case of Brazil, Fajnzylber, Maloney and Montes Rojas (2011) found that after the government implemented a policy to lower the cost of operating formally, newly created formal firms showed higher levels of revenue and profits and employed more workers. Finally, Perry (2008) found that the difference in labour productivity between firms that started up informally and firms that have always operated formally is 50 per cent in Peru, 33 per cent in Mexico, 30 per cent in Bolivia, 30 per cent in Panama and 12 per cent in Argentina. The positive productivity difference in favour of formal firms is also supported by other research at a Latin American and global level (Hsieh & Klenow 2009; Verdera 2015; Porta & Shleifer 2008).

Differences in productivity between the informal and formal sector have been estimated for Colombia. Using Matching methods, Cardenas and Mejia (2007) found significant differences in firm income per worker after controlling by observable characteristics. Along the same lines, Yvodro (2010), who used matching propensity score techniques, found that informal firms hire fewer salaried workers, mainly because they face restrictions to growth. As a result, aggregate productivity is lower. However, it is important to note that Perry et al. (2007) found no significant differences in productivity in Colombia.

Informality erodes the tax base and/or lowers the quantity and/or quality of public services and/or increases corruption

The relationship between tax revenues and informality has also been widely confirmed and explained by Levy (2008), Loayza (1997), Perry et al (2007) and Anton (2014). According to the classical model of Loayza (1997), higher tax rates generate higher tax revenues but also increase informality and therefore an optimal tax rate exists for each economy. The high correlation between VAT evasion and informality, shown in most of the literature, supports this conclusion. On the other hand, Levy (2008) argues that benefit programmes for informal workers demonstrate good intentions but end up encouraging informality, reducing productivity and diminishing government funds for social programmes.

Finally, according to De Soto (1989) informal entrepreneurs deviate 10 per cent to 15 per cent of their gross income to corruption, whereas formal entrepreneurs pay an average of only 1 per cent of gross income in bribes. This behaviour erodes the rule of law and the integrity of public institutions. Similarly, informality ends up creating unfair competition for formal firms. According to the World Bank Enterprises Survey, 28 per cent of firms worldwide report the practices of informal sector competitors as a major constraint to their functioning. This percentage amounts to an average of 31 per cent in Latin America, but is as high as 49 per cent in Brazil, 49.8 per cent in Bolivia and 55 per cent in Colombia.

Summary

In general, formal workers receive higher income, and particularly in Colombia, tend to be more satisfied with their job. The income gap appears to be positive and in favour of the formal sector for all levels of income in the case of Colombia and subject to income levels in Brazil, Chile and Mexico. These results support the findings of the previous section which

¹³ In this paper the authors define two types of workers: employees, for whom the firm pays social security (legal) and freelance contractors for whom the firm does not pay social security (informal).

point to an important segment of voluntary informality in Mexico, whereas in Colombia a combination of induced and structural informality, and to a lesser extent, voluntary informality, can be observed. The important macroeconomic cost that arises from all kinds of informality should also be stressed.

3.3. Informality and public policies aimed at improving labour market efficiency and inclusive growth: a literature review

In this section we analyse the impact on inclusive growth of public policies aimed at improving labour market efficiency, with a special focus on the Colombian tax reform of 2012 which lowered payroll tax.

Public policy analysis

The list of policies that have been implemented in different countries worldwide to improve inclusive growth through labour market efficiency is long, but can be summarised into the following two broad categories. First, policies that address the relationship between unemployment, informality and inclusive growth include: (i) promoting entrepreneurship through self-employment¹⁴; (ii) encouraging firms to formalise from inception (Klapper et al. 2007); (iii) reducing the barriers to transition from unemployment to formality (Mincer, 1991)and iv) changing unemployment benefits and subsidies (Bosch & Pretel 2013). The second category relates to policies aimed at facilitating transition from informality to formality and include: (i) reducing labour taxes (Slonimczyk 2011); (ii) reducing the minimum wage (Muravyev & Oshchepkov 2013); (iii) reducing the regulatory burden of formality (Bettcher, Friedl and Marini 2009); (iv) enforcing formality (Almedia & Carneiro 2011); (vi) establishing schemes of partial formalisation (ILO 2007), and; (vii) improving support services provided to formal firms (Campos, Goldstein & McKenzie 2015), among others.

Inevitably, analysis of these policies must take into account the specific country context. Literature indicates that for developing countries, whenever it is assumed that segmented markets exist, reducing barriers to entry into the formal sector have a marginal impact on the informal sector while regulating and taxing informality have led to increases in unemployment and poverty (Porta & Shleifer 2014). Meanwhile, analyses, whenever it is assumed more integrated markets exist conclude that reducing entry barriers (such as cutting payroll taxes) to formality is an important and useful policy to control informality. Ulyssea (2013), from a welfare analysis point of view, and assuming the coexistence of segmented and integrated markets for the case of Brazil, shows that it is possible to reduce informality by reducing entry costs to the formal sector and dropping payroll taxes. Reducing entry costs improves welfare by bringing about a substantial reduction in deadweight losses due to wasteful entry costs however the second approach does not because it cuts tax revenue. Similarly, enforcing formality in a labour market where voluntary informality exists might have a positive impact on welfare because of the new tax revenue, while enforcing formality in a market where structural informality exists might have a negative impact on welfare.

Impact of lowering payroll tax on informality

Although at a global level, the impact of lowering the payroll tax on informality has not been as widely analysed as the impact of this policy on unemployment, there are nevertheless some important findings in this field. Albrecht et al. (2009), for example, who use a theoretical model, found that payroll taxes increase informality particularly if firms are small

¹⁴ An impact evaluation conducted by CAF in 2013 showed that policies oriented to solve market failures in innovation, promote labour, entrepreneurship and skills development, and provide finance were most likely to generate higher profits and productivity among the self-employed. Meanwhile, policies targeting vulnerable entrepreneurs with low levels of education (who perceive self-employment a default option), have a lower probability of success. According to CAF, only one quarter of entrepreneurs in Latin America possesses the characteristics required to grow their firms thereby increasing employment and productivity.

and able to evade controls. Empirical works, such as Hazans (2011), have found that European countries with higher payroll taxes show higher levels of informality and Slonimczyk (2011) found that a reduction in payroll taxes in Russia had an important impact on informality indicators. As for Latin American, Lora and Fajardo (2012) found that payroll taxes increase informality if the workers do not perceive the direct benefits of these contributions, as is often the case in the region.

A number of studies analyse the relationship between payroll taxes and informality in Colombia. For example, Kugler and Kugler (2009) surveyed a panel of Colombian firms and found that an increase of 10 per cent in payroll tax was related to an increase in informal employment between 4 per cent and 5 per cent. Similarly, Mondragon et al. (2010) found that an increase in 10 per cent of payroll contributions was related to an increased probability of informality ranging between 5 and 8 percentage points. However, the impact of the 2012 reform which reduced payroll tax from 29.5 per cent to 16 per cent is yet to be documented. The exception is Anton (2014) who used a general equilibrium model to show that the 2012 tax reform should bring an increase of formal employment between 3.4 and 3.7 per cent, and a decrease in informal employment between 2.9 and 3.4 per cent.

Summary and evidence gaps

As Heintz (2012) noted, and as is still the case, very few studies analyse informality and inclusive growth at a global level. This may be because of the measurement difficulties and because of the many different forces involved in this relationship within and among the different countries.

However, the perseverance of a large informal sector in Latin America has given rise to a multitude of studies supporting the idea that informality constrains inclusive growth, both at an individual and macroeconomic level. Based on a review of the most prominent literature on these issues, we have identified several channels through which informality not only constrains but also promotes inclusive growth. In fact, some evidence suggests that informality can benefit vulnerable groups; a point of view that is rarely analysed in existing works.

The literature review also revealed considerable heterogeneity among Latin American countries with respect to the nature and behaviour of informality. For example, 'voluntary informality' appears to predominate in Mexico, whereas Colombia demonstrates a high but very heterogeneous mixture of informality. This makes the country an interesting case study because it allows us to test our two first hypotheses simultaneously. Colombia also demonstrates regional averages in terms of informality, inclusive growth and the relationship between them. Another interesting feature of Colombia is that it is one of the few countries in the region that have implemented an active policy to reduce informality. Lessons from this experience will be very useful for other countries in the region facing high formal labour costs, such as Argentina, Brazil, Costa Rica and Mexico (Graph 3.4). Notwithstanding, the impact of any policy will vary depending on the specific characteristics of the informality in each country.

In terms of specific gaps in the literature, in Latin America there is a lack of evidence around whether informality can be understood as a default option for vulnerable groups who would otherwise not be able to find a job in the formal sector. In the following sections, we attempt to fill this gap, in particular by analysing the role and relevance of informality for the older population and women heads of household. With regards to whether informality constrains inclusive growth, we will focus our attention on the impacts of formality and informality from the point of view of workers, since enterprise analysis implies the use of some problematic

databases¹⁵. Finally, although the impact of payroll tax on labour markets in Colombia has been widely analysed, little is known about the effects of the most recent tax reform which reduced payroll taxes. It is expected that analysing this policy will provide relevant insights for other regions and African countries that exhibit a combination of high payroll taxes with high informality rates.



¹⁵ Most available data is low quality and not representative at the population level.

4. Informality and inclusive growth in Colombia

In this section we present an analysis of the channels through which informality can promote and constrain inclusive growth in Colombia. We also review national policies aimed at reducing informality, in particular, the 2012 payroll tax reform.

In most of the analysis that follows, we refer to two consecutive national household surveys provided by the Department of Statistics (Dane); and the panel survey by the Universidad de los Andes. These are the Encuesta Continua de Hogares (Continuous Household Survey) (ECH 2002-2006) for urban households and the Gran Encuesta Integrada de Hogares (Widescale Integrated Household Survey) (GEIH 2007-2015) for urban households in 13 metropolitan areas, representing 60 per cent of the total urban population according to the 2005 census¹⁶. These are representative surveys; however they do not have a panel structure. In order to be able to analyse additional variables not included in the GEIH, we also refer to the Encuesta Longitudinal Colombiana de la Universidad de los Andes (Longitudinal Survey of Colombia by the Andes University) (ELCA 2010-2013). The ELCA, although not statistically representative, was gathered from around 5,000 urban households per year and was applied in a panel structure. In this chapter we mostly use the ILO5 firm definition for informality. However, when we refer to the ELCA, we classify informal workers as those who do not contribute either to state health or pension systems, the legal definition most used in Colombia. This decision is due to data limitations in estimating the firm definition. The information analysed in sections 4.1 and 4.2 of this chapter correspond to averages between April, May and June 2015, unless otherwise stated. In section 4.3, the period of analysis is adjusted in order to estimate the impact of the 2012 payroll tax reform.

4.1. Analysis of the channels through which informality might promote inclusive growth in Colombia (hypothesis 1)

In this section, we analyse whether informality acts as a substitute to unemployment in Colombia drawing on surveys, reports, figures of incidence of informality among vulnerable groups, transition matrices and analysis of counter-cyclical movements of informal employment. As is the general trend across Latin America, income gaps between unemployment and informality are not an issue in Colombia in so far as there are no sound unemployment benefits.

Drivers of informality

According to a special report of the 2006 GEIH¹⁷, 59 per cent of informal self-employed and 60 per cent of the informal salaried workers in Colombia would prefer to have a job in the formal sector at the same salary or less. Similarly, 55 per cent of self-employed workers reported the impossibility of finding a salaried job as the main reason why they were engaged in the informal sector. Table 4.1 presents these statistics by socio-economic sector according to the GEIH and similar results obtained in the ELCA¹⁸. As can be observed, all the groups that we have identified as vulnerable are likely to report this reason - inability to find a

¹⁶ The GEIH, which is conducted annually in 13 metropolitan areas, gathers information on more than 30 thousand households per month. We opted not to use the GEIH sample that covers 23 cities with rural areas since the 13 metropolitan area sample covers a longer period of time, with each month representative at a metropolitan area level. The11 remaining cities are only representative when analysed per semester. Furthermore, the 13cities sample is more frequently used by Colombian authorities

¹⁷ The 2006 report on informality is not available online. Reported by Bernal (2009) and Perry (2008).

¹⁸ Results are not totally comparable since the GEIH (2006) asks for the two main reasons and the ELCA for the main reason.

salaried job in the formal sector - as a principal cause for their self-employment status. The older population also reported 'obtaining an additional income' as an important reason for informality. These results show that, in the case of Colombia, an important portion of the informal population reports informality as being a default option to unemployment. Informality thus represents the best option for individuals whose personal characteristics, such as low education, mean they are unable to secure a job in the formal sector. However, for those individuals who meet the basic requirements for formal employment but are unable to enter into the formal sector due to high barriers, then informality may represent the second best option.

Incidence of informality among workers with low education, skills and experience In this and the next section we analyse informality among different socio-economic groups, and particularly among the most vulnerable, to explore whether informality is a default option to unemployment for them and why the formal market is not an option. A high incidence of informality among workers with low levels of education and experience who live in cities with low productivity will indicate a large incidence of subsistence informality, whereas a large incidence of informality among women and ethnic minorities will indicate that informality is the result of discrimination. Exploring this hypothesis will allow us to establish the relationship between informality and inclusive growth. Table 4.2 shows the relevant rates and shares for different socio-economic groups. This analysis also makes use of the statistics presented in Table 4.1 in order to assess the extent to which each socio-economic group enters into informality voluntarily.

Education

Education is the area where the largest differences among the socio-economic groups can be observed. The informality rate among workers with primary education or less is 79 per cent, compared with 23 per cent among workers with tertiary education. The average years of education among informal workers are 8.6 in comparison with 10.5 among the unemployed and 12.4 years among formal workers. These indicators show that an important segment of the population has such low levels of education that it is highly unlikely that they would find a job in the formal sector, which is consistent with previous results presented in this section. This finding underlines the importance of improving education as a mean to reduce informality, as was the case in Brazil (Haanwinckel et al. 2014).

On the other hand, the share of informal workers with technical and/or university studies is 17 per cent, indicating that another important segment of the population is informal despite their high level of education. The unemployment rate of this group is almost the same as the national average, probably because they are more likely to be able to afford being unemployed. This and the fact that the low, medium and high-educated population are equally represented in the unemployed group, suggests that the level of education might be more correlated with informality than with unemployment.

Table 4.1 Drivers of Informality in Colombia. Motivations of informal self-employed workers by socio-demographic characteristics (% of answers)							
			Educ	ation		Gen	nder
	Total	None	Primary	Secondary	College	Male	Female
Higher pay	9.6	4.3	7.4	11.4	17	11.5	6.6
More flexible hours	13.8	6.1	10.7	17	21.5	9	21.4
Due to his/her age	22.9	34.2	27.2	18.2	15.9	21.2	25.6
More stability or better future	1.9	0.7	1.7	2	4	2.2	1.6
Better prospects	4.1	2.3	3.6	5.2	8.6	4.6	4.6
Wishes to own his/her own firm	4.6	2.3	3.6	5.2	8.6	4.6	4.6
Less responsibility	2.4	2.2	2.5	2.5	1.6	2.2	2.6
Does not like having a boss	8.6	7.9	8.8	8.8	8.4	9.8	6.8
Family tradition	5.7	8.7	7.6	4	3	7	3.8
Inherited the business	1.3	18.8	1.7	0.9	0.3	1.7	0.8
Is used to working independently	15.6	7.9	17.9	14.1	10.2	18.2	11.5
Other reasons	10.4	7.9	10.8	20.3	52.8	7.4	12.6
Was fired has not found another job	4.7	2.6	3.5	5.6	8.3	5.2	3.9
Only job he/she could get	54.9	60.1	58	54.9	41.1	54.9	54.6
Only job he/she could get	28.7	35.7	34.1	27.6	19.9	31.6	25.9
	Δσe					Fébraia	14/2
	Less than 15	15-18	19-24*	25-44	45+**	minorities 2010	head of household
Higher pay	1.3	2.7	8.9	11.7	7.9		
More flexible hours	16.9	14	16.3	17.2	9.8		
Due to his/her age	32.2	26.5	2.3	9.1	41		
More stability or better future	0	0.6	1.3	2.5	1.5		
Better prospects	0.8	0.8	4.1	5.3	3		
Wishes to own his/her own firm	0.8	3.8	4.3	5.8	3.4		
Less responsibility	2.4	3.4	2.8	2.5	2.1		
Does not like having a boss	4.2	4.8	8.1	9.8	7.7		
Family tradition	8.1	4.4	5.3	4.8	6.8		
Inherited the business	0.2	0.8	1.4	1.3	1.5		
Is used to working independently	0.8	3.9	11.3	15.5	17.3		
Other reasons	15.7	17.3	13.2	11.1	8.8		
Was fired has not found another job	0	1.2	3.6	5.4	4.4		
Only job he/she could get	56.1	68.7	68.6	57.4	48.8		
Only job he/she could get	31.7	28.4	28.8	32.2	35.2		
	s of Informality in Higher pay More flexible hours Due to his/her age More stability or better future Better prospects Wishes to own his/her own firm Less responsibility Does not like having a boss Family tradition Inherited the business Is used to working independently Other reasons Was fired has not found another job Only job he/she could get Only job he/she could get Higher pay More flexible hours Due to his/her age More stability or better future Better prospects Wishes to own his/her own firm Less responsibility Does not like having a boss Family tradition Inherited the business Is used to working independently Other reasons Was fired has not found another job Only job he/she could get	s of Informality in Colombia. Mo Total Higher pay 9.6 More flexible hours 13.8 Due to his/her age 22.9 More stability or better future 1.9 Better prospects 4.1 Wishes to own his/her own firm 4.6 Less responsibility 2.4 Does not like having 8.6 Family tradition 5.7 Inherited the 1.3 Is used to working 15.6 Other reasons 10.4 Was fired has not found another job 4.7 Only job he/she could get 28.7 Only job he/she 28.7 Due to his/her age 32.2 More stability or better future 0 Better prospects 0.8 Wishes to own 1 his/her own firm 0.8 Less responsibility 2.4 Does not like having 32.2 More stability or better future 0 Better prospects 0.8 Wishes to own 1 his/her own firm 0.8 Less responsibility 2.4 Does not like having 4.2 Family tradition 8.1 Inherited the business 15.7 Was fired has not 0.2 Is used to working independently 0.8 Uishes to own 10.8 Less responsibility 0.8 Only job he/she 0.2 Is used to working independently 0.8 Outer reasons 15.7 Was fired has not found another job 0 Only job he/she 0.2 Is used to working independently 0.8 Outer reasons 15.7 Was fired has not found another job 0 Only job he/she could get 56.1 Only job he/she could get 31.7	s of Informality in Colombia. Motivations of inf Total None Higher pay 9.6 4.3 More flexible hours 13.8 6.1 Due to his/her age 22.9 34.2 More stability or 1.9 0.7 Better prospects 4.1 2.3 Wishes to own his/her own firm 4.6 2.3 Less responsibility 2.4 2.2 Does not like having 8.6 7.9 Family tradition 5.7 8.7 Inherited the business 10.4 7.9 Was fired has not found another job 4.7 2.6 Only job he/she 28.7 35.7 More flexible hours 16.9 14 Due to his/her age 32.2 26.5 More flexible hours 16.9 14 Due to his/her age 32.2 26.5 More flexible hours 16.9 14 Due to his/her age 32.2 26.5 More flexible hours 16.9 14 Due to his/her age 32.2 26.5 More flexible hours 16.9 14 Due to his/her age 32.2 26.5 More flexible hours 16.9 14 Due to his/her age 32.2 26.5 More stability or 0.8 3.8 Less responsibility 2.4 3.4 Does not like having 4.2 4.8 Family tradition 8.1 4.4 Inherited the business 15.7 17.3 Wishes to own 10.8 3.9 Other reasons 15.7 17.3 Was fired has not 6.2 10 Only job he/she 0.2 0.8 Wishes to work firm 1.3 1.2 Due to his/her age 3.1.7 28.4	S of Informality in Colombia. Motivations of informal self-emporation of answers of an		<t< td=""><td>Solutional self-analysis of virtual self-analysis of virtual</td></t<>	Solutional self-analysis of virtual

Source: Bernal (2009), ELCA (2013), and own calculations. In the GEIH 2007-2015, the values correspond to the percentage of options checked by respondents since the two responses per questions were permitted. The ELCA 2010-2013 asked respondents to identify the main reason for informality.

* 17-24 for ELCA ** 65+ for ELCA

Table 4.2 Rates and shares in the Colombian labour market								
		Labour mark	et ratios					
	Unemployment	Informality	Formality	Economic inactivity	Participation rate			
Total population	9.9%	48.6%	51.4%	32.0%	68%			
Women	12%	52%	48%	39%	61%			
Men	8%	46%	54%	24%	76%			
Women heads of								
household	6%	55%	45%	32%	68%			
Younger than 15	5%	92%	8%	95%	5%			
Young (15-24)	20%	42%	58%	43%	57%			
Young adult (25-35)	10%	35%	65%	11%	89%			
Young adult (36-45)	7%	46%	54%	10%	90%			
Adult (46-60)	7%	59%	41%	20%	80%			
Older than 60	4%	76%	24%	66%	34%			
Primary education or								
less	7%	79%	21%	42%	58%			
Secondary education	10%	69%	31%	51%	49%			
Higher education	11%	50%	50%	23%	77%			
Technical or university								
education	10%	23%	77%	20%	80%			
Years of education	10.5	8.6	12.4	8.1	9.7			
	Representatio	n among differei	nt socio-econor	nic groups				
	Unemployed	Informal	Formal	Economically inactive	% of total working-age population			
Women	57%	49%	43%	64%	53%			
Men	43%	51%	57%	36%	47%			
Women heads of								
household	12%	16%	13%	14%	14%			
Younger than 15	0%	1%	0%	16%	5%			
Young 15-24	36%	14%	18%	29%	21%			
Young adult (25-35)	29%	20%	35%	7%	21%			
Young adult (36-45)	14%	21%	23%	5%	16%			
Adult (46-60)	18%	32%	21%	13%	21%			
Older than 60	3%	12%	4%	30%	15%			
Primary education or								
less	13%	30%	8%	28%	21%			
Secondary education	15%	21%	9%	33%	21%			
Higher education	34%	31%	30%	19%	27%			
Technical or university								
education	38%	17%	54%	19%	31%			
Source: Own calculations based on GEIH 2007-2015								

Age – experience

Age is commonly used as a proxy for experience. According to Table 4.2, the young population¹⁹ tends to have high informality and unemployment rates (42 per cent and 20 per cent, respectively) whereas the older population has higher informality and lower unemployment (76 per cent and 4 per cent, respectively). Both groups have low participation rates. The differences in informality rates can be partially explained by the fact that the young population has studied 10.6 years on average, whereas the older population has studied an average of 6.8 years. As we suggested before, education seems to be more strongly correlated with informality than with unemployment. Another possible explication is that, according to Table 4.1, the older population tends to have stronger preferences for informal work than the younger population.



Differences between metropolitan areas

The probability of working in the formal sector very much depends on the city where one lives. While the informality rate is close to 45 per cent in big cities such as Bogotá and Medellin, it is more than 70 per cent in Cucuta, and even higher in some rural areas. Moreover, as shown in Graph 4.1, cities with a higher ratio of average wage to minimum wage demonstrate lower informality rates. This supports claims made by Mondragon et al. (2010) around the advantages of setting minimum wages closer to productivity levels. This also supports the findings of Hazans (2011) for Europe, according to which countries where minimum wages are set at a regional level how higher informality rates than those where minimum wages are fixed by central government. Based on this evidence, Fedesarrollo (2013 and 2014) has recently proposed a minimum wage by sector, as in the case in South Africa, or to return to the rural/urban minimum wages that existed in Colombia in the past. Nevertheless, the implications of this measure on income distribution are still being

¹⁹ The young population includes people between12 and 24 years old, young adults are people between 24 and 35 years of age and the older population includes individuals over 60 years old.

discussed. According to Graph 4.1, there are other forces that govern the distribution of informality among cities. For example, Pasto and Cucuta are situated close to the border and the local economy is thus impacted by smuggling. Consequently, these cities demonstrate higher levels of informality than could be expected based on productivity levels.



Incidence of informality among women and ethnic minorities

Gender

As of June 2015, the unemployment rate among women was 12 per cent and 8 per cent among men and the informality rate was 52 per cent among women and 46 per cent among men. These figures show some bias against women in the labour market in spite of the fact that the levels of education are similar between both genders. As expected, the inactivity rate is much higher among women (39 per cent) than men (24 per cent).Women are also overrepresented among the unemployed (57 per cent) and the informal (52 per cent), while they are slightly under-represented among formal workers (48 per cent) (Table 4.2).

Women heads of household are perhaps one of the most vulnerable groups in society. According to the 2005 Colombian census, women headed 25 per cent of the all households and this is a growing tendency. The unemployment rate of this group is just 6 per cent while their informality rate stands at 55 per cent; compared with national averages of 9.9 per cent and 48.6 per cent, respectively. This low rate of unemployment and high rate of informality is consistent with the high percentage of self-employed women who are head households and

who report difficulties in finding an alternative salaried job. This suggests that an important portion of this group cannot afford to be unemployed.

Ethnic minorities

In the 2005 census, around 14 per cent of the population declared itself as belonging to an ethnic minority: 10.6 per cent as Afro-Colombians and 3.4 per cent as indigenous. The GEIH does not ask questions related to ethnicity, however using the 2006 special report on informality, Bernal (2009) found that the probability of working in the informal sector is 5.4 percentage points higher for indigenous people and 2.2 percentage points higher for Afro-Colombians, controlled by other observable characteristics. Similarly, informality rates estimated using the ELCA survey suggest that the incidence of informality decreases as skin colour gets lighter, as shown in Graph 4.2. In the case of ethnic minorities there does not appear to be any specific reason for informality other than being unable to find a job in the formal sector. In fact, Bernal (2009) also found that ethnic minorities are 8 per cent more likely to prefer a formal job than the rest of the self-employed population. This is corroborated by the fact that a relatively high percentage of ethnic minorities reported being self-employed because they couldn't find a salaried job.

Summary of differences in observable characteristics

In Colombia, informal employment is over-represented among women heads of household, the older and young populations, ethnic minorities, poorly educated individuals and border city workers. If all individuals with at least two of the previous characteristics are categorised as 'vulnerable'²⁰, this group would account for 11 per cent of the working-age urban population. The high rate of informality of this group, 78 per cent, cannot be explained by specific high preferences for self-employment, since we observed that they reported even higher preferences for salaried work than the rest of the population²¹. Furthermore, informality on this specific group does not appear to be caused by barriers to formality. In fact, given low productivity levels amongst vulnerable groups²², it would be very difficult for these workers to find a job in the formal market even if formal market restrictions were drastically reduced. Taking all this evidence into account, we can assert that in Colombia informality is a default option to unemployment and promotes inclusive growth at an individual level among vulnerable groups.

Using the information gathered in this section, we also estimated the probability of formality. As shown in Table 4.3, all the coefficients obtained from the logit demonstrate the expected signs. The groups that we have identified as vulnerable demonstrate low odds ratios, which means that they have lower probabilities of working in the formal sector, whereas men, household heads, young adults, workers with higher education and inhabitants of Bogotá, Medellin and Cali are more likely to work in the formal sector. According to Table 4.1, this high tier group also reported higher preferences for informal jobs related to factors such as higher pay, more flexible hours and being their own boss.

²⁰ Excluding ethnic minorities, if we classify as 'vulnerable' all those individuals who show at least one of these characteristics the size of the group would amount to half of the national population and their informality rate would be 58 per cent.

²¹ With the only exception of the older population.

²² The average earnings of this vulnerable population groups are 33 per cent lower than the average earnings of the informal worker and 30 per cent lower than the minimum wage.

Table 4.3								
Logit mode	I. Dependent	variable: pro	bability of	formality				
	Odds ratio	Std error	Z	Sig.				
	1.5	0.0	388	***				
Head of household	1.3	0.0	218	***				
Women Head of household	1.0	0.0	-12	**				
15-24 years	1.2	0.0	162	***				
25-35 years	1.4	0.0	299	***				
45-60 years	0.7	0.0	-329	***				
60+ years	0.4	0.0	-550	***				
Less than primary	0.2	0.0	-301	***				
Primary	0.5	0.0	-646	***				
Higher	2.3	0.0	530	***				
Technical or university	2.4	0.0	508	***				
Big city	1.6	0.0	506	***				
Border city	0.6	0.0	-222	***				
Constant	0.4	0.0	-695	***				
LR chi2(13)	7507820							
Prob> chi2	0							
Pseudo R2	0.1697							

How frequent are transitions between groups? Is informality a buffer to unemployment and/or a step towards formality?

Another characteristic which allows us to identify whether informal workers are integrated with the unemployed or with formal workers is to analyse the frequency of transitions between groups. In doing so, we use the ELCA survey which provides panel data for 2010 and 2013. According to Table 4.4, the probability of remaining informal in 2013 for workers who were informal in 2010 is 72 per cent. For formal workers in 2010, the probability of remaining in formal employment in 2013 is also 72 per cent. This shows a high degree of persistence of both informality and formality in Colombia. For several reasons, including methodological ones, our calculations show that people are less likely to remain unemployed. Table 4.4 shows that informality does not necessarily lead on to a formal job because the probability of transitioning to formal employment from informal employment is just 14 per cent; lower than the probability of transitioning to formality from unemployment (19 per cent). However, informality might be a buffer to unemployment because the probability of transitioning from formal to informal employment is 20 per cent, which is higher than the probability of transitioning from formal to informal employment to unemployment (3 per cent).

We also analysed transition matrices for different segments of the population. As we can see in Table 4.5, the persistence of men in the formal sector is higher than persistence in the informal sector for men. The same is true for the highly educated population and young adults. In contrast, the persistence of women and poorly educated workers is higher in informality than in formal jobs. In the case of the older population persistence in the formal and informal sectors is the same at 72 per cent, which is the same persistence rate of the population as a whole. Table 4.5 also shows that persistence in any segment of the market unemployment, informality or formality - is higher for men than for women. In other words, men tend to transit less than women across different statuses. It is also possible to observe in Table 4.5 that the probability of transitioning from informality to formality is higher for men than women, higher for young and young adults than for the older population, and higher for highly educated workers than for workers with primary education or less. In contrast, the probability of transitioning from formality to informality is higher for women, for older workers and for less educated workers. In every group of workers, however, the probability of transitioning from formal employment to unemployment is much lower than probability of moving from formal to informal employment.

			2013		
		Unemployed	Informal	Formal	Inactive
010	Unemployed	18%	35%	19%	27%
	Informal	4%	72%	14%	10%
Ñ	Formal	3%	20%	72%	6%
	Inactive	4%	26%	3%	67%

Table 4.5 Transition matrices by gender 2013 2013 Female Male Formal Unemployed Informal Inactive **Unemployed Informal** Formal Inactive 2010 Unempl 14% 38% 17% 31% 2010 Unemployed 27% 30% 24% Informa 6% 68% 9% 17% Informal 3% 75% 17% Formal 22% 17% 77% 3% 65% 9% Formal 3% Inactive 5% 26% 3% 66% Inactive 2% 24% 7% Older population (60+) Young and young adults (15-35) Unemployed Informal **Unemployed Informal** Formal Formal Inactive Inactive 2010 2010 Unempl 18% Unemployed 13% 35% 19% 27% 36% 20% Informa 4% 72% 10% Informal 4% 20% 14% 68% Formal 3% 20% 72% 6% Formal 5% 16% 76% Inactive 4% 26% 3% 67% Inactive 10% 40% 5% Tertiary education Elementary education or less Unemployed Informal **Unemployed Informal** Inactive Formal Inactive Formal 2010 2010 Unempl 9% 30% Unemployed 14% 20% 40% 11% 30% 12% 26% Informa 5% 73% 10% Informal 2% 66% 2% 31% 60% 7% Formal Formal 1% 12% 83% Inactive 4% 21% 1% 74% Inactive 2% 30% 11% Source: Own calculations based on ELCA 2013

Transitions in and out the labour force are also interesting. According to the ELCA, 26 per cent of the inactive population in 2010 entered the labour force through informality in 2013, and 27 per cent of the unemployed population became inactive in 2013. The former is mostly the result of young individuals becoming active, but unable to find a formal job. The latter could be the result of people getting tired of looking for a job or it taking too long to find employment, which is consistent with high numbers of individuals with higher education becoming unemployed.

19%

5%

3%

68%

31%

8%

4%

44%

45%

6%

4%

57%

The main conclusion here is that informality promotes inclusive growth insofar as it acts as a buffer to unemployment, yet at the same time it can also constrain inclusive growth due to the difficulty of moving from informal to formal sector jobs. It was also found that there are asymmetries in these transitions, with vulnerable groups more prone to transit from formality to informality, and less likely to use informality as a step towards finding a formal job. This analysis assumes that the transition from informality of formality promotes inclusive growth, which is the subject of discussion in subsequent sections.

Is informality a buffer to unemployment?

The counter-cyclicality of informality is not only important as a benefit of informality itself but also because it is very much related to the question of how voluntary or involuntary informal employment is and therefore whether informality promotes or constrains inclusive growth. The left-hand panel of Graph 4.3 shows that there is a positive relationship between the formality rate²³ and the economic cycle, measured as the relative difference between observed and potential GDP. The correlation coefficient between the formality rate and the output gap is 0.74 for the ILO10 series and 0.74 for the ILO5 series until 2013. These results support the hypothesis pertaining to the counter-cyclicality of informal employment in Colombia.

The positive relationship between the economic cycle and the formality rate does not necessarily imply causality between economic growth and formality. It can be argued that high rates of economic growth can be a consequence of low informality. In order to isolate this type of double causation, we plotted the relationship between the formality rate and the value of commodity exports as a percentage of GDP trend in the right-hand side of Graph 4.3. Commodity exports represent a good proxy for the economic cycle since they are exogenous to informality and well correlated with the output gap²⁴. The correlation coefficient between formality and commodity exports is 0.73 for the ILO10 series and 0.63 for the ILO5 series. Therefore, we can claim that the formality rate in Colombia is in general pro-cyclical and the informality rate is in general counter-cyclical, which is consistent with having a significant portion of involuntary informal workers amongst whom informality can increase inclusive growth. It also supports the idea of informality being a buffer to unemployment during crisis.

However, as shown in Graph 4.3, the most recent years show an important increase in formality rates that cannot be explained by the economic cycle. In fact, the correlation coefficient between ILO5 and both measures of the output gap, drop to 0.43if we include the years 2014 and 2015. As we will later demonstrate, this might be related with the 2012 payroll tax reform. In other words, a relaxation of formal market rigidities might have resulted in more pro-cyclical behaviour of the informal sector.

Is informal employment a substitute to unemployment?

Based on evidence reviewed, informality appears to represent a substitute to unemployment for the cyclically and structurally unemployed. In other words, informality represents the only alternative option for vulnerable groups and it is difficult to argue against the idea that informality promotes inclusive growth through this channel. However, for the segment of the population with at least secondary school studies, informality may represent a constraint rather than a step towards inclusive growth. The following section explores these issues further.

²³ Defined as one minus the informality rate.Note that the formality rate is calculated for two different ILO methodologies/series since one includes firms with less than 10 workers (ILO10 2010) and the other includes firms with less than 5 workers (ILO5).

²⁴ See Fernandez, Villar & Sánchez (2015).



Analysis of the channels through which informality might constrain inclusive growth in Colombia (hypothesis 2)

To evaluate whether informality constrains inclusive growth by impacting growth and jobs in the formal sector, we analyse data from household surveys which allow us to explore whether informal employment implies lower productivity and generates lower income, benefits and stability than formal employment. Other channels that we have identified, but shall not investigate here, relate to whether: informality generates lower labour productivity according to firms; informality lowers tax revenues and/or the quantity and/or quality of public services, and; informality increases corruption. We have opted not to explore these channels further since they are already well documented in existing literature and require different sets of data than the ones we use here.

Does informal employment generate lower income, benefits and stability than formal employment?

Average income of formal workers is almost double the average income of informal workers in Colombia. However, this gap might be related to differences in individual characteristics, such as education, gender and city of work, amongst others, as demonstrated in the previous section. With the caveats explained in the literature review²⁵, we calculated a Matching model for comparing workers' income controlled by observable characteristics, with the formal sector as a treatment group and informal workers as the control group. As shown in Table 4.6, the average income of the formal sector before the matching was two times the average

²⁵

Wage differences between formal and informal workers should be taken with care when used because they can reflect differences in preferences for informal work or other unobservable variables, which can be partially controlled by dividing the sample in socio-economic groups.

income in the informal sector, however once controlled by observable characteristics, or matched, the wage gap lowered to 1.5 times²⁶. Table 4.7 shows the test that confirms the quality of the matching²⁷.

Table 4.6 Differences in income, stability and satisfaction in Colombia before and after adjusting by observable characteristics									
	uajat	Formal	Informal	Ratio	T-stat				
Labour income	Unmatched	1308440	644807	2.0	49.4				
	Matched	1308440	846798	1.5	30.4				
Stability	Unmatched	80%	66%	1.2	30.8				
	Matched	80%	67%	1.2	20.4				
Work satisfaction	Unmatched	78%	68%	1.1	21.4				
	Matched	78%	67%	1.2	18.2				
Health benefits	Unmatched	97%	90%	1.1	28.2				
	Matched	97%	88%	1.1	21.7				
Differences in labour income per group									
Formal Informal Ratio T-stat									
Primary									
education or less	Unmatched	780023	516648	1.5	19.9				
	ATT*	780023	607136	1.3	13.0				
Secondary									
education	Unmatched	858356	672035	1.3	15.7				
	ATT	858466	713981	1.2	11.5				
Tertiary education or									
more	Unmatched	1703256	913491	1.9	19.5				
	ATT	1703413	972054	1.8	23.1				
Older population	Unmatched	1673901	646982	2.6	33.4				
	ATT	1673901	960047	1.7	17.2				
Women Head of	Unmatched	1369093	519175	2.6	28.1				
Household	ATT	1369093	765452	1.8	15.7				

Source: Own calculations, based on GEIH first quarter 2015

* Average treatment on the treated, the effect of the programme on the treatment group minus the situation of the treatment group if the individuals of the treatment group had not been participants of the programme.

²⁶ The logit model and the p-score that we used to do the matching are the same as those used in the previous section. The method used to match the p-scores of the informal workers with those of the formal workers is the kernel method – Epanechnikov - which has the advantage of using most of the observations and reducing the variance.Leuven and Sianesi (2003) developed the Stata code used in this exercise.

We tested the quality of matching according to Rosenbaum and Rubin (1985), which stipulates that the standardised bias should be less than 5 per cent. Although the bias was large for the unmatched co-variables (U in Table 4.6), after matching (M) all the control variables complied with this criterion. The average standard bias that is what really matters since we are working with p-scores is only 1.6 per cent.

The lower panel of Table 4.6 presents the unmatched and the adjusted comparisons in income between formal and informal workers by education, gender and age. Performing this disaggregated matching helps to reduce the bias generated by unobservable characteristics. We find that income is always higher for formal than for informal workers. However, the differences between unmatched and adjusted income are higher for women and the older population, as well as for the low educated group. This last result confirms findings by Herrera et al. (2013) and Nuñez (2002) regarding the disadvantages of having lower levels of education in the Colombian labour market.

	Table 4.7								
		Fitting	the mat	cning m	odel (probability of formality)				
Variable		Test	Control	%bias (std)	ed_superior				
Male	U	56%	50%	11.60	diploma				
	М	56%	56%	-1.40	edad3				
Head of	U	44%	48%	-7.10					
Household	М	44%	44%	-0.40					
Women Head of	U	13%	18%	-14.20	jefe				
Household	Μ	13%	13%	0.20	ciudadfrontera				
15-24 years	U	18%	13%	12.30					
	Μ	18%	19%	-2.50	ed_0				
25-35 years	U	33%	19%	32.00	edad6				
	Μ	33%	31%	5.70	ed_prim				
45-60 years	U	22%	32%	-24.40					
	M	22%	23%	-3.10	-50 0 50 100 Standardized % bias across covariates				
60+ years	U	4%	13%	-33.40					
	M	4%	4%	0.50	Unmatched				
	U	0%	3%	-21.90					
Less than primary	Μ	0%	0%	0.30	Q -				
Primary	U	7%	28%	-57.30	15. 15.				
	M	/%	/%	0.70	Ö				
Secondary	U	55%	16%	86.70	ଞ -				
Technical or	IVI	55%	54%	1.30					
university	U	46%	12%	81 70	Standardized % bias across covariates				
university	M	46%	45%	2.70	Matched				
Big city	U	36%	27%	21.20	<u>ଶ୍</u> ୟୁ -				
	M	36%	36%	0.60	∾				
Border city	U	10%	14%	-13.70	<u>5</u> 2				
,	М	10%	9%	1.20					
Average (p-score)	U			32.10	8				
	М			1.60					
			0.5.11.2		-88 -66 -44 -22 0 22 44 66 88 Standardized % bias across covariates				
Source: Own calc	ulatio	ons, based	on GEIH 2	007-2015					

Nevertheless, income might not be the only reason to look for a formal job. For example, formal jobs tend to be more stable than informal jobs. As can be observed in Table 4.6, 80 per cent of formal workers consider that their job is stable compared to 66 per cent of informal workers. These numbers are quite similar after adjusting by observable characteristics. Similarly, other benefits might provide work satisfaction other than income and stability, such as flexibility and the satisfaction of being the own boss. As the results

show, 78 per cent of formal workers are satisfied with their jobs whereas the rate of satisfaction is only 67 per cent among informal workers, after adjusting by observable characteristics. Table 4.6 also shows that informal workers have high rates of health coverage at 88 per cent, compared with 97 per cent of formal workers. This does not necessarily mean that informal workers are contributing to the health system, since they may receive benefit from the subsidised scheme. This result suggests that the size and quality of the welfare state in Colombia plays an important mediating role in the effect of informality on inequality with the formal sector.

In sum, we find a positive income gap in favour of formal employment. However, as stated in the literature review, this difference should be analysed with care since it does not necessarily imply that workers in the formal sector are better-off than informal workers. The gap can also imply differences in unobservable characteristics or preferences for formal/informal work. Analysing the results by groups of individuals can reduce, though not remove, this discrepancy. In general, formal works also provide more satisfaction, health coverage and stability than informal jobs.

Is informal employment associated with lower labour productivity than formal employment?

It is possible to analyse productivity differences between formal and informal firms with the household surveys by using labour income as a proxy for labour productivity. We replicated the previous exercise controlling not only by observable characteristics but also by firm size. We also applied the definition of informality that measures health contributions in order to avoid including firm size, the new control variable, in the definition of the treatment group. As shown in Table 4.8, the productivity ratio between formal and informal firms is 1.5, after controlling for firm size and worker observable characteristics²⁸.

Table 4.8 Differences in labour income* after adjusting by observable characteristics and size							
		Formal	Informal	Ratio	T-stat		
Controlling for size	Unmatched	1,323,776	576,290	2.3	56.0		
only	ATT	1,323,776	709,176	1.9	28.6		
Controlling for size	Unmatched	1,323,776	576,290	2.3	56.0		
and observable characteristics	ATT	1,314,642	858,766	1.5	20.0		

Source: Own calculations, based on GEIH first quarter of 2015 *Understood as marginal productivity

Summary

Formal jobs seem to generate higher income, satisfaction and stability than informal employment. In particular, much higher wage differentials are observable for vulnerable groups in the formal and the informal sectors. However, these findings should be interpreted with care given the difficulty of controlling for unobservable characteristics. The results also show that informal workers, particularly more vulnerable groups, receive less income and are not as protected against shocks, which implies a constraint on inclusive growth.

²⁸

Unfortunately the average mean grows up to 6.5, which is on the higher side. Changing the specification, we obtained the same results and a better matching, however we preferred to maintain the co-variables used in all the exercises for consistency.

Given the fact that formal firms are generally bigger than informal firms, these results can be partially explained by size. However, we repeated the exercise and found that the conclusions remain the same after controlling for size, which also provides some evidence in favour of productivity differences between formal and informal firms. This reinforces what we learned from the literature review in terms of the macroeconomic costs of informality, such as tax evasion, unfair competition and higher corruption. It should be kept in mind that although informality can be positive in terms of inclusive growth for some vulnerable groups, the cost of informality at a society level should not be ignored. These costs might only become visible when informality accounts for a large portion of the working population and is more difficult to control.

4.2. Policy analysis in Colombia: the impact of lowering payroll tax (hypothesis 3)

The Colombian government recently reformed the tax law by reducing payroll contributions from 29.5 per cent to 16 per cent and substituting them with a profit tax. This substitution only affects the payments made by the employers of workers that earn wages between one and ten times the minimum wage, and does not change the amount of taxes or contributions payable by the workers. Passing and implementing the law involved several milestones. Most of the discussions were held between October and November 2012, the law was approved in December 2012, some of the reductions in contributions became effective in May 2013 and the reform was fully implemented on 1st January 2014.

The period following the reform coincided with an important reduction in informality rates. However, it also coincided with high economic growth rates and a significant reduction in unemployment rates. It is therefore of great interest to investigate whether the recent reduction in informality, from 51 per cent in 2012 to 48.3 per cent in 2015, was due to high rates of growth or to the tax reform; and therefore, whether informality will continue to drop in the current climate of low economic growth, which has resulted from a reduction in export commodity prices and a general economic deceleration among South American economies.

The impact of payroll tax reform on informality

One of the most adequate methodologies for isolating the impact of growth over time is the Differences in Differences method. Slonimczyk (2011) uses this technique to analyse the impact of a tax reform in Russia. The method involves dividing the population into two groups: one affected by the reform, the *treatment* group, and the other unaffected by the reform, the *control* group. The change in probability of informality within the control group is then compared with the change observed in the probability of informality within the treatment group. By taking the difference between these changes –or the difference in differences- one isolates factors that affect both groups simultaneously, such as macroeconomic conditions, assuming that the impact on informality is evenly spread between both groups. As Todd (1999) claims, the advantage of this methodology compared to a cross-section analysis is that it allows for time-invariant unobservable differences between the treatment and the control groups.

As a first step, we performed this exercise for the case of Colombia, using the following equation and Ordinary Least Squares (OLS)

$$INF_{it} = \theta_t + X_{it}\beta + \psi Time_t + \mu Treated_i + \alpha (Treatment_i \times Time_t) + u_{it}$$

where X_{it} refers to the observable characteristics, Time is a dummy variable that takes the value of zero in the baseline period and a value of one in the period after the reform, and Treatment is a dummy that takes the value of one if the individual is from the treatment group and zero if not. The treatment group includes all workers that were directly impacted by the

reduction in payroll taxes²⁹. The control group includes all other workers. In our baseline estimations, we excluded self-employed and public sector employees in order to reduce differences between the control and treatment groups in terms of unobservable characteristics, but we relaxed this constraint later. The control variables are the same as those used in previous sections. However, we included dummies per month in recognition of the existence of some seasonality.

Graph 4.4 plots the informality rate for the treatment and control groups before controlling by observable characteristics. As can be observed, although the reform was discussed in the second semester of 2012 and approved in December 2012, it only started to have an impact in January 2014 when it was fully implemented. After this period, the model confirms that workers affected by the reform, or the treatment group, were less likely to engage in the informal sector, while this was not the case for workers in the control group. Graph 4.4 also indicates that relatively long-term moving averages should be considered in this type of analysis since the series demonstrates considerable volatility. Therefore, we defined our period of analysis from the first semester of 2012, before the implementation of the reform, to the first semester of 2014, after the implementation of the reform. Second semesters of both years were analysed separately.

The results of the OLS exercise are presented in Table 4.9 which shows the impact of the tax reform on the treatment group was statistically significant, having reduced informality by 6.5 percentage points; whereas among the control group the informality rate increased by 3.7 percentage points. All the control variables show the expected signs. However, this exercise has some limitations because it assumes common time effects across groups³⁰ and no changes to the composition of each group. These assumptions are less easy to control when there is no panel structure (Blundell 2009).

To partially reduce these limitations, it is possible to use a procedure called 'Matching Differences in Differences' developed by Heckman et al. (1997). Villa (2011) developed the Stata code that we used in this paper. The idea is to match the treatment individuals after the reform with treatment individuals before the reform and with the control individuals before and after the reform, and then compare the differences in informality rates between the treatment and control groups over time³¹.

According to the law, this includes workers that earn between one and 10 times the minimum wage excluding NGO workers. We also included those workers who reported an income close enough to the minimum wage or to ten times the minimum wage. In fact, we realised that a number of formal workers who probably earn the minimum wage rounded this figure up to the next ten thousand Colombian Pesos, and therefore we included them in the treatment group.

³⁰ In fact, the model can control for non-observable individual specific effects and non-observable macroeconomic effects because they cancel one another out, but not for non-observable temporary individual specific effects.

³¹ By mixing the methodologies, Matching and Differences in Differences, the assumption of common time effects on unobservables becomes common trend time effects on un-observables. This method does not require the linear assumption (Blundell 2009).



Table 4.9 Results of the differences in differences methods (OLS)						
	Coef.	Std. Err.	t	p>t		
January	0.02	0.00	6.3	0.0		
Male	0.00	0.00	0.0	1.0		
Head household	0.00	0.00	1.0	0.3		
Women Head of Household	0.00	0.01	0.4	0.7		
15-24 years	-0.01	0.00	-3.1	0.0		
2-54-35 years	-0.03	0.00	-8.5	0.0		
45-60 years	0.04	0.00	9.0	0.0		
60+ years	0.17	0.01	21.1	0.0		
Less than primary	0.07	0.02	4.4	0.0		
Primary	0.03	0.01	5.7	0.0		
, Secondary	-0.12	0.00	-25.8	0.0		
Higher education	-0.20	0.01	-32.4	0.0		
Technical or university	-0.05	0.01	-8.5	0.0		
Big city	-0.04	0.00	-12.3	0.0		
Border city	0.04	0.00	7.8	0.0		
Test	-0.28	0.00	-58.8	0.0		
Year	0.04	0.01	6.4	0.0		
Test*year	-0.07	0.01	-9.8	0.0		
_cons	0.67	0.01	104.1	0.0		
Number of obs	89058					
F(18, 89039)	1121.27					
Prob> F	0					
R-squared	0.1848					
Adj R-squared	0.1846					
Root MSE	0.42176					
Source: Own calculations, based on GEIH 2007-2015 *Times						

Table 4.10 Impact of the lowering of payroll taxes Results from the differences in differences and matching methods								
Baseline (excludes self-employment)								
First semester 2014/First semester 2012								
		Baselin	e	F	ollow up			
	Control	Test	Difference	Control	Test Difference		DID	
Informal workers	52.20%	25.40%	-26.90%	55.90%	21.70%	-34.20%	-7.40%	
Std. Error	0.003	0.003	0.004	0.003	0.003	0.004	0.006	
т	168.16	81.7	-61.13	180.82	70.16	-78.25	-11.86	
P> t	0	0	0	0	0	0	0.00%	
	-	Seco	ond semester 2014/S	Second semester 2012				
		Baselin	e	F	ollow up			
-	Control	Test	Difference	Control	Test	Difference	DID	
Informal workers	53.40%	25.00%	-28.40%	57.20%	20.90%	-36.30%	-8.00%	
Std. Error	0.003	0.003	0.004	0.003	0.003	0.004	0.006	
т	173.8	81.5	-65.3	188.7	68.8	-84.7	-13	
P> t	0	0	0	0	0	0	0.00%	
			Quality of th	e matching				
Weighted Variable(s)	Mean	Control	% Bias (std)	Weighted Variable(s)	Mean	Control	% Bias (std)	
Less than primary	0.60%	0.60%	0	Male	59.10%	59.20%	-0.2	
Primary	12.40%	12.10%	0.92	Head of household	47.40%	47.40%	0	
Secondary	35.50%	34.50%	2.1	Women Head of Household	11.40%	11.60%	-0.63	
Higher education	40.10%	41.10%	-2.04	15-24 years	16.00%	16.10%	-0.27	
Technical or university	31.90%	32.80%	-1.92	25-35 years	33.40%	34.80%	-2.95	
Month 1	15.90%	16.10%	-0.5	45-60 years	22.80%	21.30%	3.62	
Month 2	16.30%	16.00%	0.8	60+ years	3.40%	2.70%	4.07	
Month 3	17.30%	17.30%	0	Big city	37.00%	37.00%	0	
Month 4	17.40%	17.50%	-0.3	Border city	8.80%	8.60%	0.09	
Month5			0.4	Total			0.09	
	-		Placeb	o test				
		F	irst semester 2011/I	First semester 2013				
Baseline Follow up								
	Control	Test	Difference	Control	Test	Difference	DID	
Informality rate	56.1%	25.9%	-30.2%	53.3%	23.8%	-29.5%	-0.7%	
P> t	0%	0%	0%	0%	0%	0%	76%	
		F	irst semester 2011/I	First semester 2013				
Informality rates	56%	26%	-30%	53%	24%	-29%	1%	
P> t	0%	0%	0%	0%	0%	0%	17%	

Table 4.10 shows the results of applying the Matching and Difference in Difference methods for the first semester of 2014³². According to the results, informality rates among the treatment group reduced by 7.4 percentage points in the analysis period due to the reduction in payroll taxes or to a shock impacting the control but not the treatment group, such as the reduction in payroll taxes. We also performed the exercise including self-employment and the results show a reduction in the informality rate of the treatment group of 6.8 percentage points³³. Similar results were also obtained when analysing the impact between the second semester of 2012 and the second semester of 2014 when the informality rate of the treatment group reduced by 7.9 percentage points due to the reduction in payroll tax or any similar shock. These outcomes are comparable to what Anton (2014) predicted for the tax reform using a general equilibrium model and what Kugler and Kugler (2009) and Mondragon et al. (2010) predicted having examined historic changes in payroll tax.

Some interesting conclusions were also found in analysing the specific transitions of the treatment and control groups during the period of analysis (2012-2014). In fact, while the informality rate of the treatment group in comparison with the control group decreased approximately 7.7 percentage points due to the tax reform, some of this change was explained by formal workers earning less than a minimum wage becoming fully formal workers earning at least a minimum wage and probably making pension and health contributions³⁴.We did not find the same behaviour around the upper bound of the reform i.e. ten times the minimum wage.

In order to test the quality of the matching, we applied the criteria developed by Rosenbaum and Ruben (1985)³⁵. Table 4.8 shows that after the matching, all variables and the aggregate bias, which is what matters the most since we are working with p-scores - fulfil this criterion. We also implemented a Placebo Test to check the common trend assumption, according to which unobservable variables such as growth, affect the outcome variable, informality, of treatment and control groups in an equal way. For the Placebo Test, the Matching and Differences in Differences Methods are applied to any other year with similar external characteristics, faking the existence of a tax reform or a similar shock, with the expectation that the results will not be affected. We performed this exercise in relation to the previous period - the first and second semesters of 2012 - and compared the results with the first and second semesters of 2013, a period which showed similar external characteristics in terms of

³² In estimating the p-score, we employed the same logit model used throughout the research and we included a dummy variable per month to control for seasonality. When conducting the matching methodology, we used the kernel procedure developed by Epachnikov, which has the advantage of reducing variance and making use of most available information.

³³ Not including self- employment in the estimations, the fulfillment of the parallel trends assumption for a considerable period of time before the reform was weaker; given particular changes in the self-employment affected the control but not the treatment group in 2010. This was one of the reasons for excluding self-employment from the main exercise. It is very likely that change in trends in 2010 was the result of incentives to increase formalisation at that time, particularly directed towards the self-employment. The behavior of both groups after these incentives is relatively stable and the results of the exercise including and not including self-employment are similar, suggesting that the exercise including self-employment is not heavily biased.

³⁴ By assuming that the formal workers were responsible for all the increase in the treatment group, we estimated that at most 4.1 of this change was due to this type of endogenity. It should also be noted that the aim of the reform was to obtain exactly the same results that were obtained by the increase in the treatment group i.e. to fully formalise the working population.We also found that the change in the treatment group was overestimated by the survey, meaning that if we use the Matching Differences in Differences procedure using weights we will find a lower but more robust impact of the reform.

³⁵ We tested the quality of matching by using Rosenbaum and Rubin's (1985) rule of thumb according to which the standardised bias should be less than 5 per cent. Table 4.8 shows that, after matching, all the control variables complied with this criterion and the average bias in the p-score - that is actually what matters the most in the estimation - is 0.4 per cent. The standardised bias for the dummy co-variables was estimated as:

SB= 100 *($p_c - p_t$) / [{ $p_t (1 - p_t) + p_c (1 - p_c)$ } * 1/2] ^{1/2}, where p_c is the proportion of each co-variable in the control group and p_t is the proportion of each co-variable in the treatment group.

growth but did not witness major changes in labour policy³⁶. We obtained no significant results in informality between the treatment and control groups, as shown in Table 4.10. Although this test does not exclude the violation of the common trend assumption, it does suggest that growth or other unobserved variables did not impact informality in a differential manner³⁷.



Impact of the payroll tax reform on income distribution

In order to be able to apply lessons from the Colombian case to other contexts and to analyse the impact of the reform on income distribution, we explored the characteristics of the workers most affected by the reform. We began by analysing the behaviour of the informality rate per income quintile for the first semester of 2012 and the first semester of 2014.We performed the exercise for both observed labour income and imputed labour

³⁶ Same results were obtained between the first and second semesters of 2011 and the first and second semesters of 2013.

³⁷ We had some concerns with the individuals who represent the formal workers in the control group in terms of the unobservable characteristics. This group includes some formal workers who earn more than 10 times the minimum wage, however most formal worker earn less than a minimum wage. This is not very common in a country like Colombia where the minimum wage prevails in the formal sector. Therefore, it might be the case of these workers having different unobservable characteristics when comparing them to the informal workers in the control group. However, as we are working with Differences in Differences this bias is netted when subtracting the pre-reform results, unless these unobservable characteristics cause un-even changes in the rate of informality during the observed period. We did not find any reason for such an impact, as the placebo test confirms.

income³⁸. As shown in Graph 4.5, informality lowered during the period of analysis primarily amongst the middle-income quintile which includes minimum wage earners³⁹. When we performed the Differences in Differences exercise per socio-economic group, shown in Table 4.11, we also found that the workers with secondary education or less benefited most from the reform. This can be explained by the fact that the reform removed a constraint that was bigger for minimum wage earners compared to workers receiving higher levels of income where wages are more flexible.

Table 4.11 Impact of the tax reform by population group									
	Outcomo	Baseline		Follow up		DID			
Low Educated (Primary or less)	Informality P > t P > t	72% 0%	47% 0%	77% 0%	41% 0%	-10.14% 0%			
Secondary	Informality P > t P > t	55% 0%	26% 0%	64% 0%	23% 0%	-11.9% 0%			
Tertiary education or higher	Informality P > t P > t	37% 0%	14% 0%	40% 0%	13% 0%	-4.26% 0%			
Male	Informality P > t P > t	54% 0%	28% 0%	58% 0%	24% 0%	-8.21% 0%			
25-35 years	Informality P > t P > t	48% 0%	18% 0%	52% 0%	16% 0%	-6.61% 0%			
60+ years	Informality P > t P > t	72% 0%	55% 0%	72% 0%	49% 0%	-4.9% 13%			
Women head of household	Informality P > t P > t	52% 0%	26% 0%	53% 0%	23% 0%	-4.0% 3%			

Summary

The tax reform had a relatively significant impact on the informality rate of the treatment group after controlling for observable and some unobservable characteristics. This result reflects the most common findings in other studies. However, the reduction in informality seems to have been a one-off improvement and therefore new increases can be expected as a result of the current stage in the economic cycle. We also found that the payroll tax reform had the most impact on the middle-income population because workers in this group earn close to the minimum wage where the constraint was released.

³⁸ Using a Tobit function and years of education, gender, age, squared age and city as independent variables. The fit is reasonably accurate but it does not show the high density at the minimum wage.

³⁹ Second or fourth quintile, depending whether we used the current or imputed income.

5. Main findings

Does informality promote or constrain inclusive growth? After reviewing relevant literature for Latin America and conducting a detailed analysis of the case of Colombia, our answer to this question is: it depends. As Levy (2008) claimed, the problem is not informality *per se* rather it is to *too much* informality which can constrain inclusive growth. In a context where it is important to promote decent working conditions yet formal market restrictions cannot be totally removed, some informality has to be tolerated. In fact, informality plays an important role in the economy by providing employment to a segment of the population that otherwise would not find a job in the formal market. However, too much informality constrains inclusive growth, particularly when it is caused by excessive labour market restrictions which prevent workers from operating in a more productive environment and from obtaining higher wages, benefits and work stability. Likewise, informality constrains inclusive growth when it is pursued on the basis of a voluntary decision, which generates negative externalities at a society level. More specific conclusions can be summarised as follows.

Informality is a major problem in Latin America and particularly in Colombia, however the informality rate has diminished recently.

The average informality rate in Latin America is high at 46 per cent. In countries such as Bolivia, Honduras and Paraguay informality is as high as 70 per cent. Colombia has an average informality rate of 48 per cent. Over the last decade, informality rates have declined across the region and in Colombia, probably due to high economic growth. However, there is a chance that informality may rise again as a result of recent downward trends in growth and considering that in most countries of the region, informality demonstrated counter-cyclical behaviour.

Latin America has a poor record of inclusive growth. Nonetheless, recent growth was accompanied by an impressive reduction in poverty. This is also the case in Colombia

Latin America is a region with medium levels of inclusive growth. Colombia sits in the medium- lower range and demonstrates one of the most negative relationships between inclusive growth and informality. In recent years, economic growth has been accompanied by remarkable results in poverty reduction in Latin America and particularly in Colombia where poverty rates almost halved from 50 per cent in 2002 to 28.5 per cent in 2014. However, there is some uncertainty concerning whether this trend will be resilient in the current down turn in growth.

Informal workers in Latin America, and particularly in Colombia, are a dynamic mixture of diverse groups which demonstrate different relationships between informality and inclusive growth

Amongst vulnerable groups in Latin America, high informality rates are accompanied by relatively high preferences for salaried work. These findings lead us to the conclusion that for these groups, informality is indeed a default option to unemployment. It is difficult to argue against informality being an alternative that has increased inclusive growth for this group. However, for the segment of the population with at least secondary school studies, informality may represent a constraint rather than a step towards inclusive growth, especially considering the high persistence rates of informality which can be considered a sort of poverty trap for the region. In general, the informal sector in Latin America is extremely diverse, with Mexico showing a higher rate of 'voluntary' informality and the informal sector in Colombia demonstrating greater heterogeneity. These findings go in line with Garcia (2014) who identifies heterogeneity within the Colombian informal sector, whereby some informal workers desire a formal job whereas a relatively smaller proportion chooses to be informal.

Nevertheless, the cost of informality at a society level, which is greater when informality includes a bigger proportion of the working population, cannot be ignored

It is also important to note that although informality can be positive in terms of inclusive growth for some vulnerable groups, the costs at a society level should not be ignored. In fact, there are significant productivity differences between formal and informal sector firms and this reinforces what we learned from the literature in terms of other macroeconomic costs of informality such as tax evasion, unfair competition and higher corruption.

The payroll tax reform in Colombia produced a one-time significant but moderate reduction in informality, and generated some positive effects on income distribution

Our calculations show the Colombian 2012 tax reform had a relatively high impact of about 7 percentage points on the informality rate among workers targeted by the reform after controlling for observable and some unobservable characteristics. While some of this increase was the result of formal workers with flexible work becoming fully formal workers who earn a minimum wage, this result is very much in line with the most common findings of previous studies. On the other hand, this reduction seems to have been a one-off improvement and therefore new increases in informality can be expected as a result of the current stage in the economic cycle. We also found that the payroll tax reform impacted mostly the middle-income population. This can be explained by the fact that a greater constraint was released for minimum wage level earners compared to higher level income groups where wages are more flexible.

6. Conclusions and policy recommendations

When we started this research, we expected to find evidence that informality constrains inclusive growth, as common wisdom in the region dictates. However, as our analysis proceeded we discovered that informality plays an important role in society, particularly for vulnerable groups. This is a rather innovative perspective in the Latin American context.

One of the main conclusions in this paper is that informality cannot be analysed based on the assumption that informal workers are a bulk mass of individuals with similar characteristics. Informal workers in Colombia range from poorly educated individuals, which may be classified as structurally informal, to highly educated young adults living in productive cities who can be included in the voluntary informality group. These groups show very different relationships between informality and inclusive growth.

This analysis has important policy implications. Although an important bulk of the informal population is affected by formal employment barriers, 'structural informality' needs to be tackled with other kind of policies such as education. In the case of voluntary informality, imposing constraints on any remaining informal and economic incentives to become informal might be effective, whereas the same policies applied to structural informality might have a negative impact on inclusive growth. This lesson can be applied to most countries in Latin America and Africa. However, it is important to note that even if informality can be positive in terms of inclusive growth for some vulnerable groups, the cost of informality at a society level should not be ignored.

Regarding the recent Colombian payroll tax reform, our estimations show a one-time significant but moderate reduction in the informality rate of 7 percentage points among the target groups after controlling for observable characteristics and partially controlling for unobservable characteristics. Some of this increase was the result of formal workers with flexible work becoming fully formal workers who earn a minimum wage. This experience is of the most interest for countries with high levels of payroll taxes on top of high and enforced minimum wages.

References

- Alcaraz, C.; Chiquiar, D. and Salcedo, A. (2015) *Informality and Segmentation in the Mexican Labor Market,* Working Papers. No 2015-25, Mexico City: Banco de México
- Ali, I. and Son, H.H. (2007) Measuring Inclusive Growth, Asian Development Review 24.1: 11
- Almeida, R. and Carneiro, P. (2011) *Enforcement of Labor Regulation and Informality,* IZA DP 5902, Bonn: Institute for the Study of Labor
- Almeida Ramos, R.; Ranieri, R. and Lammes, J.W. (2013) *Mapping Inclusive Growth,* International Policy Centre for Inclusive Growth Working Paper 105, Brasilia: International Policy Centre for Inclusive Growth
- Anand, R.; Mishra, M.S. and Peiris, S.J. (2013) Inclusive Growth: Measurement and Determinants, International Monetary Fund Working Paper 13-135, Washington DC: IMF
- Antón, A. (2014) 'The Effect of Payroll Taxes on Employment and Wages Under High Labor Informality', *IZA Journal of Labor and Development* 3.20
- Antón, A. and Hernández, F. (2010) VAT Collections and Social Security Contributions: Is There a Link?, Washington DC: Inter-American Development Bank
- Arias, O. and Bustelo, M. (2007) *Profiles and Dynamics of Informal Employment in Latin America*, Washington DC: World Bank
- Arias, O., Landa, F. and Yáñez, P. (2007) *Movilidad Laboral e Ingresos en el Sector Formal e Informal de Bolivia [Work and Income Mobility in the Formal and Informal Sector Bolivia],* Documento de Trabajo, La Paz: UDAPE
- Badel, A. and Peña, X. (2010) *Decomposing the Gender Wage Gap with Sample Selection Adjustment: Evidence from Colombia,* Working Papers 045, Federal Reserve Bank of St Louis
- Bargain, O. (2010) 'Is Informality Bad? Evidence from Brazil, Mexico and South Africa', *Review of Income and Wealth* 57: 100–22
- Bassi, M.; Busso, M.; Urzúa, S. and Vargas, J. (2012) *Desconectados: Habilidades, Educación y Empleo en América Latina [Disconnected: Skills, Education and Employment in Latin America],* Washington DC: Inter-American Development Bank Education Unit
- Bernal, R. (2009) 'The Informal Labour Market in Colombia: Identification and Characterization', *Desarrollo y Sociedad* 63: 145–208
- Bettcher, K.; Friedl, M. and Marini, G. (2009) *From the Streets to Markets: Formalization of Street Vendors in Metropolitan Lima*, Case Study 0901, Washington DC: Center for International Private Enterprise
- Bosch, M. and Maloney, W.F. (2008) *Cyclical Movements in Unemployment and Informality in Developing Countries,* World Bank Policy Research Working Paper Series, Washington DC: World Bank

- Bosch, M. and Pretel, J. (2013) *The Labor Market Effects of Introducing Unemployment Benefits in an Economy with High Informality,* IBD Working Paper Series 402, Washington DC: Inter-American Development Bank Labor Markets and Social Security Unit
- Busso, M.; Fazio, M.V. and Levy, S. (2012) (In)Formal and (Un)Productive: The Productivity Costs of Excessive Informality in Mexico, IBD Working Paper Series 341, Washington DC: Inter-American Development Bank Department of Research and Chief Economist
- Campos, F.; Goldstein, M. and McKenzie, D. (2015) *Short-Term Impacts of Formalization* Assistance and a Bank Information Session on Business Registration and Access to Finance in Malawi, Policy Research Working Paper 7183, Washington DC: World Bank
- Cardenas, M. and Mejia, C. (2007) 'Informalidad en Colombia: Nueva Evidencia [Informality in Colombia: New Evidence]', *Cuadernos de Fedesarrollo* 35
- Cassim, A.; Lilenstein K.; Oosthuizen M. and Steenkamp F. (2015) *Informality and Inclusive Growth in Sub-Saharan Africa. Evidence and Lessons from Latin America (ELLA) Regional Evidence Paper,* Lima: Practical Action Latin America
- Cord, L.; Genoni, M.E. and Rodriguez Castelan, C. (2015) *Shared Prosperity and Poverty Eradication in Latin America and the Caribbean*, Washington DC: International Bank for Reconstruction and Development/World Bank
- Dabla, N.; Gradstein, M. and Inchauste, G. (2005) *What Causes Firms to Hide Output? The Determinants of Informality,* International Monetary Fund Working Paper 05-160, Washington DC: IMF
- Dabla-Norris, E. and Koeda, J. (2008) *Informality and Bank Credit: Evidence from Firm-Level Data,* International Monetary Fund Working Paper 08-94, Washington DC: IMF
- Daza, N. and Gamboa, L.F. (2013) *Informal-formal Wage Gaps in Colombia,* ECINEQ Working Paper Series 2013-301, Palma de Mallorca: Society for the Study of Economic Inequality
- De Brauw, A.; Gilligan, D.O.; Hoddinott, J. and Roy, S. (2014) 'The Impact of Bolsa Família on Women's Decision-making Power', *World Development* 59: 487–504
- De Holanda Barbosa, A.L.N. and Corseuil, C.H.L. (2014) 'Cash Transfer and Informality in Brazil', *IZA Journal of Labor and Development* 3.1: 1–18
- De Soto, H. (2000) *The Mystery of Capital: Why Capitalism Triumphs in the West and Fails Everywhere,* Berkshire: Black Swan
- Economic Commission for Latin America and the Caribbean (ECLAC) and the International Labour Organization ILO (2014) *The Employment Situation in Latin America and The Caribbean: Employment Formalization and Labour Income Distribution*, Santiago de Chile: United Nations
- ECLAC, ILO (2015) Panorama Temático Laboral: Transición a la Formalidad en América Latina y el Caribe, Lima: ILO Regional Office for Latin America and the Caribbean
- El Badaoui, E. and Rebière, T. (2012) *Informal Labor Market and Access to Education in Developing Economies,* Laboratoire d'Econométrie and IZA, Department of Economics, Paris: CNAM

- Fajnzylber, P.; Maloney, W. and Rojas, G.M. (2006) 'Microenterprise Dynamics in Developing Countries: How Similar are they to those in the Industrialized World? Evidence from Mexico', *The World Bank Economic Review* 20.3: 389–419
- Fedesarrolllo (2014) Algunas Consideraciones sobre el Mercado Laboral Regional en Colombia [Some Considerations on the Regional Labor Market in Colombia], Informe Mensual del Mercado Laboral, Noviembre del 2014, Bogotá: Fedesarrollo
- Fedesarrollo (2013) El Debate Sobre el Salario Mínimo Regional [The Debate on Regional Minimum Wage], Informe Mensual del Mercado Laboral, Marzo del 2013, Bogotá: Fedesarrollo
- Fedesarrollo, Fundación Saldarriaga Concha (2015) *Misión Colombia Envejece: Cifras, Retos y Recomendaciones [Colombia is Aging Mission: Numbers, Challenges and Recommendations],* Bogotá: Editorial Fundación Saldarriaga Concha
- Fernández, C.; Villar, L. and Sánchez, P.M. (2015) ¿Hay Vida Después de las Bonanzas? [Is there Life after Bonanzas?], Fedesarrollo Working Paper 68, Bogotá: Fedesarrollo
- Fiess, N.; Fugazza, M. and Maloney, W.F. (2008) *Informality and Macroeconomic Fluctuations*, IZA Discussion Paper 3519, Bonn: IZA
- FORLAC (2015) Promoting Formal Employment Among Youth: Innovative Experiences in Latin America and the Caribbean, Lima: ILO Regional Office for Latin America and the Caribbean
- FORLAC (2014) Recent Experiences of Formalization in Latin America and the Caribbean. Notes on Formalization, Lima: ILO Regional Office for Latin America and the Caribbean
- Funkhouser, E. (1997) 'Mobility and Labor Market Segmentation: The Urban Labor Market in El Salvador', *Economic Development and Cultural Change* 46.1: 123–53
- Galli, R. and Kucera, D. (2008) *Gender, Informality and Employment Adjustment in Latin America,* Working Paper 85, Geneva: ILO
- García, G. (2014) Labor Informality: Choice or Sign of Segmentation? A Quintile Regression Approach at the Regional Level for Colombia, Archivos de Economía. Documento 412, Bogotá: National Planning Department, Government of Colombia
- Gong, X. and Van Soest, A. (2002) 'Wage Differentials and Mobility in the Urban Labor Market: A Panel Data Analysis for Mexico', *Labour Economics* 9.4: 513–29
- Gong, X.; Van Soest, A. and Villagomez E. (2004) 'Mobility in the Urban Labor Market: A Panel Data Analysis for Mexico', *Economic Development and Cultural Change* 53.1: 1–36
- Haanwinckel, D. and Soares, R. (2013) A Compensating Differentials Theory of Informal Labor Markets: Quantitative Model and Implications for a Developing Country, LACEA conference paper
- Harris, J.R. and Todaro, M.P. (1970) 'Migration, Unemployment, and Development: A Two-Sector Analysis', *The American Economic Review* 60.1: 126–42
- Hazans, M. (2011) Informal Workers Across Europe: Evidence From 30 Countries, Discussion Paper Series IZA 5871, Bonn: IZA

- Heintz, J. (2012) Informality, Inclusiveness, and Economic Growth: An Overview of Key Issues, Supporting Inclusive Growth (SIG) Programme, International Development Research Centre (IDRC) Working Paper, Ontario: IDRC
- Henley, A.; Arabsheibani, G.R. and Carneiro, F.G. (2006) On Defining and Measuring the Informal Sector (Vol. 2473), Washington DC: World Bank
- Herrera-Idárraga, P.; López-Bazo, E. and Motellón Corral, E. (2013) *Double Penalty in Returns to Education: Informality and Educational Mismatch in the Colombian Labour Market*, IREA–Working Paper 2013/0, Barcelona: IREA
- Hsieh, C. and Klenow, P. (2009) 'Misallocation and Manufacturing TFP in China and India', *Quarterly Journal of Economics* 124.4: 1403–48
- ILO (2013a) Panorama Laboral (Labour Panorama), Geneva: ILO
- ILO (2013b) Thematic Labour Overview 1. Transition to Formality in Latin America and the Caribbean, Lima: ILO, Regional Office for Latin America and the Caribbean
- ILO (2007) *The Informal Economy: Enabling Transition to Formalization,* Tripartite Interregional Symposium on the Informal Economy: Enabling Transition to Formalization, Geneva: ILO
- India Central Statistical Organisation (1999) Report of the Third Meeting (New Delhi, 17–19 May 1999), Expert Group on Informal Sector Statistics (Delhi Group), New Delhi: Central Statistical Organisation
- Klapper, L.; Amit, A.; Guillen, M. and Quesada, M. (2007) *Entrepreneurship and Firm Formation Across Contries,* Policy Research Working Paper 4313, Washington DC: World Bank
- Kugler, A. and Kugler, M. (2009) Labor Market Effects of Payroll Taxes in Developing Countries: Evidence from Colombia, Chicago: University of Chicago
- Leuven X.E. and Sianesi B. (2003) *PSMATCH2: Stata Module to Perform Full Mahalanobis* and Propensity Score Matching, Common Support Graphing, and Covariate Imbalance Testing. Online publication accessed March 2016
- Levy, S. (2008) Good Intentions, Bad Outcomes: Social Policy, Informality, and Economic Growth in Mexico, Washington DC: Brookings Institution Press
- Lewis, W.A. (1954) 'Economic Development with Unlimited Supplies of Labour,' *The Manchester School* 22.2: 139–91
- Loayza, N. (1997) 'The Economics of the Informal Sector: A Simple Model and Some Empirical Evidence from Latin America', *World Bank Policy Research Paper* 1727, Washington DC.: World Bank
- Loayza, N. and Rigolini, J. (2006) 'Informality Trends and Cycles', *World Bank Policy Research Working Paper* 4078, Washington DC: World Bank
- Lora, E.A. and Fajardo, D.J. (2012) *Employment and Taxes in Latin America: An Empirical* Study of the Effects of Payroll, Corporate Income and Value-added Taxes on Labor

Outcomes, IBD Working Paper Series 334, Washington DC: Inter-American Development Bank Department of Research and Chief Economist

Maloney, W.F. (2004) 'Informality Revisited', World Development 32.7: 1159-78

- Maurizio, R. (2015) 'Labour Economy in Latin America. The Case of Argentina, Brazil, Chile and Peru', in Cling, J.P., Lagrée, S., Razafindrakoto, M., Roubaud, F. Routledge, *The Informal Economy in Developing Countries,* London and New York: Routledge Taylor & Francis Group
- Maurizio, R. (2013) Income Distribution and Labour Market in Latin America in Times of Economic Growth, Buenos Aires: Universidad Nacional de General Sarmiento
- McKinley, T. (2010) *Inclusive Growth Criteria and Indicators: An Inclusive Growth Index for Diagnosis of Country Progress,* Asian Development Bank, Sustainable Development Working Paper 14, Manilla: ADB
- Mincer, J. (1991) *Education and Unemployment,* NBER Working Paper 3838, Cambridge: National Bureau of Economic Research
- Mondragon-Velez, C.; Peña, X.; Wills, D. and Kugler, A. (2010) 'Labor Market Rigidities and Informality in Colombia', *Economía*, 65–101
- Muravyev, A. and Oshchepkov, A. (2013) *Minimum Wages, Unemployment and Informality: Evidence from Panel Data on Russian Regions,* IZA Discussion Paper 7878, Bonn: IZA
- Nuñez, J.A. (2002) Empleo Informal y Evasión Fiscal en Colombia [Informal Employment and Tax Evasion in Colombia], Bogotá: National Planning Department, Government of Colombia
- Pacchioni, E.A.G. (2013) Andemic Informality: Assessing Labor Informality, Employment and Income Risk in the Andes, Washington DC: Inter-American Development Bank
- Peña, X. (2013) The Formal and Informal Sectors in Colombia, Geneva: ILO
- Perry, G.E.; Maloney, W.F.; Arias,O.S.; Fajnzylber, P.; Mason, A.D. and Saavedra-Chanduvi, J. (2007) *Informality: Exit and Exclusion*, Latin American and Caribbean Studies, Washington DC: World Bank
- Porta, R.L. and Shleifer, A. (2014) *Informality and Development,* NBER w20205, Cambridge: National Bureau of Economic Research
- Porta, R.L. and Shleifer, A. (2008) *The Unofficial Economy and Economic Development,* NBER Working Paper w14520, Cambridge: National Bureau of Economic Research
- Pratap, S. and Quintin, E. (2006) *The Informal Sector in Developing Countries,* Helsinki: United Nations University
- Rosenbaum, P.R. and Rubin, D.B. (1985) 'Constructing a Control Group Using Multivariate Matched Sampling Methods that Incorporate the Propensity Score', *The American Statistician* 39.1: 33–8
- Slonimczyk, F. (2011) The Effect of Taxation on Informal Unemployment. Evidence from the Russian Flat Tax Reform, London: London School of Economics

- Teixeira, C.G. (2010) A Heterogeneity Analysis of the Bolsa Família Programme Effect on Men and Women's Work Supply, Working Paper 61, Washington DC: International Policy Centre for Inclusive Growth
- Todd, P. (1999) A Practical Guide to Implementing Matching Estimators, guide prepared for IADB Meeting, Santiago de Chile
- Tokman, V. (2008) 'Informality in Latin America: Facts and Opportunities', Background Paper for the Swedish International Development Cooperation Agency (Sida), unpublished Working Paper
- Tornarolli, L.; Battistón, D.; Gasparini, L. and Gluzmann, P. (2014) *Exploring Trends in Labor Informality in Latin America, 1990-2010*, La Plata: CEDLAS Universidad Nacional de la Plata
- Ulyssea, G. (2013) Firms, Informality and Welfare, Chicago: University of Chicago
- Villa, J.M. (2011) *DIFF: Stata Module to Perform Differences in Differences Estimation. Statistical Software Components*, Boston: Boston College
- Villar, P. (2013) Formalización: Motor de Cambio Estructural Positivo. El Caso Colombiano Entre 2002 y 2011 [Formalization: A Positive Motor Structural Change. The Colombian Case Between 2002 and 2011], Bogotá: Universidad de los Andes
- Sakho, Y.S.; Lunde, T. and Arribas-Banos, M. (2009) *Gender in Bolivian Production: Reducing Differences in Formality and Productivity of Firms,* World Bank country study, Washington DC: World Bank

World Bank (2015) 2015 World Development Indicators, Washington DC: World Bank

- World Bank (2014) 2014 World Development Indicators, Washington DC: World Bank
- World Bank Enterprise Surveys, <u>www.enterprisesurveys.org</u>, Washington DC: World Bank
- Ydrovo, C. (2010) 'Informalidad Empresarial en Colombia: Un Obstáculo a la Destrucción Creativa [Business Informality in Colombia: An Obstacle to Creative Destruction]', mimeo, Bogotá: Banco de la Republica